Qualitative spatial reasoning: A new approach to cyclic ordering of 2D orientations^{*}

Amar Isli

WSV, Fachbereich Informatik, Universität Hamburg, Vogt-Kölln-Strasse 30, D-22527 Hamburg, Germany isli@informatik.uni-hamburg.de

Anthony G. Cohn School of Computer Studies, University of Leeds, Leeds LS2 9JT, United Kingdom agc@scs.leeds.ac.uk

March 31, 2000

Abstract

We propose a new approach to cyclic ordering of 2D orientations, consisting of a relation algebra (RA) whose universe is a set of ternary relations. An atom of the RA expresses for triples (z_1, z_2, z_3) of orientations whether each of the three orientations is equal to, to the left of, opposite to, or to the right of each of the other two orientations. The RA has 24 atoms and the elements of its universe consist of all possible 2^{24} subsets of the set of all atoms. Because we are dealing with ternary relations, we add *rotation* as an operation in addition to those present in Tarski's formalisation of RAs. Amongst other results, (1) we provide for the RA a constraint propagation procedure computing the closure of a problem under the different operations, which we show is polynomial, and complete for a subset including all atoms; (2) we prove that another subset, expressing only information on parallel orientations, is NP-complete; (3) we show that provided that a subset S of the RA includes two specific elements, deciding consistency for a problem expressed in the closure of

^{*}A preliminary version of this paper, prepared while the first author was at the University of Leeds, has appeared in the Proceedings of the Fifteenth American Conference on Artificial Intelligence, pages 643-649, AAAI Press, 1998.

 $^{^\}dagger$ This work was supported by the Spatial Inference project of the DFG priority program on Spatial Cognition, under grant Fr 806/7-2, and by EPSRC under grants GR/K65041 and GR/M56807.

S can be polynomially reduced to deciding consistency for a problem expressed in S; and (4) we derive from the previous result that we "jump" from tractability to intractability if we add the universal relation to the set of all atoms of the RA. A comparison to the most closely related work in the literature indicates that the approach is promising.

Keywords: Qualitative spatial reasoning, Relation algebra, Constraint satisfaction, Orientation, Computational complexity, Knowledge representation.

1 Introduction

Qualitative spatial reasoning (QSR) has become an important and challenging research area of Artificial Intelligence. An important aspect of it is topological reasoning (see the survey in [5]). However, many applications (among which are robot navigation [22], reasoning about shape [34], route description [7, 17]) require the representation and processing of orientation knowledge. A variety of approaches to this have been proposed: the so-called CYCORD theory for cyclic ordering of 2D orientations [27, 31, 32], global reference system models for reasoning about cardinal directions [8, 9], relative orientation models [11, 12, 18, 40], and models for the representation of a panorama [33].

One may want to describe a configuration of points in the plane as viewed from a global point of view; this could, for instance, correspond to the situation when a robot has to be located with respect to a number of known landmarks: such a description may consist of specifying the cyclic order of triples of objects in the configuration with respect to the viewpoint at the robot's location. The CYCORD theory [31, 32] and Schlieder's system of panorama representation [33] may be used for such a task. However, in addition to providing the cyclic order for triples of orientations, many applications may need the specification for pairs of orientations in the configuration of whether one orientation is to left of, to the right of, opposite to, or equal to, the other orientation. This feature is not captured by the system in [31, 32] nor by the one in [33]: indeed, these neglect what could be called the *left/straight/right* dichotomy determined by an observer placed at the point of view and looking in the direction of the reference object; this dichotomy, which corresponds to the partitioning of the plane into the directed line point-of-view-reference-object, the left open half-plane delimited by that line, and the right open half-plane delimited by the same line, allows, when captured by a model, for some kind of cognitively plausible reasoning (some aspects of cognitive plausibility of orientation models in qualitative spatial reasoning are discussed in [11, 12]).

To illustrate, consider the simple natural language sentence "You see both the university and the hill on your left when you walk down to the station": the CYCORD theory fails to provide a representation of this description. Another limitation of the CYCORD theory appears when we consider the *samedirection/opposite-direction* dichotomy determined by the same observer referred

to above, which splits the directed line point-of-view-reference-object into the positive part, i.e., the part the observer is looking at, and the negative part, i.e., the part at the back of the observer. This second dichotomy is also important for qualitative spatial reasoning applications, as illustrated by the descriptions "The cinema is on the way to the university", or "To get to the cinema from the station, walk in the opposite direction to the university". This motivates the need for a new, finer grained, approach to cyclic ordering of 2D orientations, which is what we propose in the paper. The new approach, which is an atomic relation algebra (RA) whose universe is a set of ternary relations on 2D orientations, overcomes the above limitations; furthermore, as it turns out, its atoms form a tractable subset, which is important for at least two reasons:

- 1. Complete information can be checked for consistency in polynomial time.
- 2. Deciding consistency for a general problem expressed in the RA, which we show is NP-complete, can be achieved using a backtracking search procedure, which refines at each node of the search tree the relation on a triple of 'variables' to an atom.

The RA represents knowledge on cyclic ordering of 2D orientations as a ternary constraint satisfaction problem (ternary CSP) of which:

- 1. the variables range over the set of 2D orientations, which, as we will see later, is isomorphic to the set of points of a fixed circle, as well as to the set of directed lines containing a fixed point; and
- 2. the constraints give for triples of the variables the relation of the RA they should satisfy.

We first define a binary RA and, based on that, develop our new approach to cyclic ordering. Among other things, we provide a composition table for the binary RA. One reason for doing this first is that it will then become easy to understand how the relations of the ternary RA are obtained.

The binary RA can model the qualitative configuration of two orientations. It is based on the left/straight/right and same-direction/opposite-direction dichotomies mentioned earlier, both determined by an observer placed at the point of view and looking in the direction of the reference object. The point of view, say P, is global, and we make the realistic assumption that if a collection of point objects is to be qualitatively described relative to P then all objects in the collection are different from P. The point of view may, for instance, be a robot and the objects in the collection landmarks: equality of the position of the robot and that of one of the landmarks would correspond to a collision! In this way, given two objects A and B, it makes sense to consider the orientations z_1 and z_2 of the directed lines (PA) and (PB), respectively, which can be qualitatively compared according to the two dichotomies mentioned above: z_2 is e(qual) to, to the l(eft) of, o(pposite) to, or to the r(ight) of, z_1 . To



Figure 1: Localisation of a robot R with respect to four landmarks L1, L2, L3, L4.

illustrate, consider the situation in Figure 1 where a robot R has to be qualitatively localised relative to four landmarks L_1, L_2, L_3, L_4 . This can be achieved by considering the orientations Z_1, Z_2, Z_3, Z_4 of the directed lines $(RL_1), (RL_2),$ $(RL_3), (RL_4)$, respectively, joining the robot to the landmarks. We can then use the binary RA to represent the situation as a description specifying the relation holding on each pair of the four orientations. For instance, to "the robot is to the right of the directed line (L_1L_2) " corresponds the relation $r(Z_2, Z_1)$, stating that orientation Z_2 is to the right of orientation Z_1 .

So far, constraint-based approaches to QSR have mainly used constraint propagation methods achieving path consistency. These methods have been borrowed from qualitative temporal reasoning à la Allen [1], and make use of a composition table. It is, for instance, well-known from works of van Beek that path consistency achieves global consistency for CSPs of Allen's convex relations. The proof of this result, given in [36, 37], shows that it is mainly due to the 1-dimensional nature of the temporal domain. The proof uses the specialisation of Helly's theorem [4] to n = 1: "If S is a set of convex regions of the *n*-dimensional space \mathbb{R}^n such that every n+1 elements in S have a non empty intersection then the intersection of all elements in S is non empty". For the 2-dimensional space (n = 2), the application of the theorem gets a bit more complicated, since one has to check non emptiness of the intersection of every three elements, instead of just every two; we will use this to show that a constraint propagation procedure to be given for the ternary RA, which achieves strong 4-consistency, has a similar behaviour for a subset including all atoms as path consistency for Allen's convex relations: the procedure achieves global consistency.

The paper is organised as follows. Section 3 provides some background on constraint satisfaction problems (CSPs) and constraint matrices. Section 4 provides some background on relation algebras (RAs), and focusses on binary RAs and ternary RAs. Section 5 presents our first RA of 2D orientations,

 \mathcal{CYC}_b , which is binary. The second RA, \mathcal{CYC}_t , which is ternary, is presented in detail in Section 6. We then come back, in Section 7, to CSPs with a focus on CSPs of 2D orientations: \mathcal{CYC}_b -CSPs, i.e., CSPs of which the constraints are \mathcal{CYC}_b relations on pairs of the variables; and \mathcal{CYC}_t -CSPs, i.e., CSPs of which the constraints are \mathcal{CYC}_t relations on triples of the variables. The section provides an example showing that path consistency is not sufficient for deciding consistency for a CSP of \mathcal{CYC}_b atoms; then a constraint propagation procedure achieving strong 4-consistency for \mathcal{CYC}_t -CSPs, which we show is polynomial; and finally, a procedure to search for a strongly 4-consistent, thus consistent, scenario of a general \mathcal{CYC}_t -CSP. In Section 8, we show that a subset of \mathcal{CYC}_t including all atoms is tractable; specifically, we show that our strong 4-consistency procedure can decide consistency for a CSP expressed in the subset. In Section 9, we present some intractability results:

- 1. From the NP-completeness of the CYCORD theory [15], we derive that \mathcal{CVC}_t is NP-complete.
- 2. We show that a subset of \mathcal{CVC}_t expressing only information on parallel orientations is NP-complete.
- 3. We show that provided that a subset S of CYC_t includes two specific elements, deciding consistency for a CSP expressed in the closure of S under the different CYC_t operations can be polynomially reduced to deciding consistency for a CSP expressed in S.
- 4. From the previous result, we derive that for both RAs the set obtained by adding the universal relation to the set of all atoms is NP-complete.

Section 10 compares our approach to the most closely related ones in the literature. Directions for future work are discussed in Section 11. Finally, Section 12 summarises the paper.

We first need to motivate the use of ternary relations.

2 Motivation of the use of ternary relations

We make the assumption that the 2D space is associated with a reference system (O, x, y), and refer to the circle centred at O and of unit radius as $C_{O,1}$, and to the set of 2D orientations as 2DO. Three natural isomorphisms will be of use in the rest of the paper. In order to facilitate their definitions, we introduce the following sets:

- 1. $\mathcal{R}_{0,1}$ is the set of all radii of $\mathcal{C}_{O,1}$ excluding the centre O but closed at the other endpoint.
- 2. $d\mathcal{L}_O$ is the set of all directed lines containing O.
 - 5



Figure 2: The angle (D_1, D_2) determined by two directed lines D_1 and D_2 is the one corresponding to the move in an anticlockwise direction from D_1 to D_2 .

Definition 1 The isomorphisms $f_1^{2DO\rightarrow}$, $f_2^{2DO\rightarrow}$ and $f_3^{2DO\rightarrow}$ are defined as follows:

$$f_1^{2D\mathcal{O}\to}: 2D\mathcal{O} \to \mathcal{R}_{0,1} \tag{1}$$

 $f_1^{\mathcal{2}D\mathcal{O}\rightarrow}(z)$ is the radius $(OP_z] \in \mathcal{R}_{0,1}$ such that the orientation of the vector $\overrightarrow{OP_z}$ is z.

$$f_2^{2D\mathcal{O}\to}: 2D\mathcal{O} \to \mathcal{C}_{O,1} \tag{2}$$

 $f_2^{\mathcal{D}\mathcal{O}\rightarrow}(z)$ is the point $P_z \in \mathcal{C}_{O,1}$ such that the orientation of the vector $\overrightarrow{OP_z}$ is z.

$$f_3^{\mathcal{D}\mathcal{D}\mathcal{O}\to}:\mathcal{D}\mathcal{O}\to d\mathcal{L}_O \tag{3}$$

 $f_3^{\mathcal{DO} \to}(z)$ is the line $\ell_{O,z} \in d\mathcal{L}_O$ of orientation z.

Definition 2 The angle determined by two directed lines D_1 and D_2 , denoted (D_1, D_2) , is the one corresponding to the move in an anticlockwise direction from D_1 to D_2 (see Figure 2). The angle (z_1, z_2) determined by orientations z_1 and z_2 is the angle $(\ell_{O,z_1}, \ell_{O,z_2})$, where $\ell_{O,z_1} = f_3^{2DO \to}(z_1)$ and $\ell_{O,z_2} = f_3^{2DO \to}(z_2)$.

The set 2D \mathcal{O} can thus be viewed as the set of radii of $\mathcal{C}_{O,1}$ (or, indeed, of any fixed circle), as the set of points of $\mathcal{C}_{O,1}$ (or of any fixed circle), or as the set of directed lines containing O (or any fixed point). We will not restrict ourselves to any of these sets; however:

- 1. in order to illustrate the relation holding between 2D orientations, it seems more intuitive to look at orientations as directed lines containing a fixed point, for instance O (isomorphism $f_3^{2D\mathcal{O}\rightarrow}$). For example, an orientation z_1 is to the left of an orientation z_2 if the angle (z_2, z_1) belongs to $(0, \pi)$; and
- 2. for the proof of Theorem 6, we will look at an orientation as a radius of $\mathcal{C}_{O,1}$ excluding the centre O (isomorphism $f_1^{2D\mathcal{O}\rightarrow}$).

The aim of this work, as stated in the introduction, is to provide a relation algebra (RA) for representing and reasoning about cyclic ordering of 2D orientations. For qualitative ordering of elements of a linear universe, such as the standard time line, it is sufficient to use a binary relation over the universe; however, ordering elements of a cyclic universe, such as the universe of 2D orientations (or any of the isomorphic universes $\mathcal{R}_{0,1}$, $\mathcal{C}_{O,1}$ and $d\mathcal{L}_O$), requires the use of a relation of arity at least equal to three. Together with a convention saying which of the anticlockwise and clockwise directions corresponds to the positive direction, a ternary relation with the aim of qualitatively ordering 2D orientations would give for triples (z_1, z_2, z_3) of orientations which of z_2 and z_3 is met first when we move in the positive direction starting from z_1 . This shows the importance of ternary relations for our purpose. On the other hand, many applications, including those mentioned in the introduction, may require finer knowledge on triples (z_1, z_2, z_3) of orientations than just the (qualitative) cyclic order of z_1 , z_2 and z_3 : for instance, they may require for some or all pairs (z_i, z_j) of orientations in $\{z_1, z_2, z_3\}$ the additional knowledge of whether z_i is equal to, to the left of, opposite to, or to the right of, z_i . As alluded to briefly in the introduction, the partitioning of the universe of 2D orientations into the orientation that is equal to, the orientations that are to the left of, the orientation that is opposite to, and the orientations that are to the right of, a given orientation (parent orientation) is important because of its cognitive plausibility [11, 12]: the parent orientation might correspond to the orientation of the directed line (PR) joining a parent object P to a reference object R; then an observer placed at the parent object and looking in the direction of the reference object can describe a primary object S relative to the reference object R in the following, cognitively plausible, way, where $z_{(PR)}$ and $z_{(PS)}$ stand for the orientations of the directed lines (PR) and (PS), respectively: S may be in front of the observer, colinear with P and R $(equal(z_{(PS)}, z_{(PR)}))$; to the left of R $(left(z_{(PS)}, z_{(PR)}))$; at the back of the observer, collinear with P and R $(opposite(z_{(PS)}, z_{(PR)}));$ or to the right of R $(right(z_{(PS)}, z_{(PR)}))$. Thus what is needed is to combine a cyclic ordering, thus ternary, relation with these other, binary, relations, equal, left, right, and opposite, in order to offer the possibility of expressing finer grained knowledge than just cyclic ordering. This paper provides a calculus to satisfy this need.

3 Constraint satisfaction problems

A constraint satisfaction problem (CSP) of order n consists of a finite set of n variables x_1, \ldots, x_n , a set U (called the universe of the problem), together with a set of constraints on values from U which may be assigned to the variables. The problem is solvable if the constraints can be satisfied by some assignment of values $a_1, \ldots, a_n \in U$ to the variables x_1, \ldots, x_n , in which case the sequence (a_1, \ldots, a_n) is called a solution. Two problems are equivalent if they have the

same set of solutions.

An *m*-ary constraint is of the form $R(x_{i_1}, \dots, x_{i_m})$, and asserts that the *m*-tuple of values assigned to the variables x_{i_1}, \dots, x_{i_m} must lie in the *m*-ary relation R (an *m*-ary relation over the universe U is any subset of U^m). An *m*-ary CSP is one of which the constraints are *m*-ary constraints. We will be concerned exclusively with binary CSPs and ternary CSPs.

A unary relation, say R, is equivalent to the binary relation $\{(a, a) : a \in R\}$, and to the ternary relation $\{(a, a, a) : a \in R\}$. In turn, a binary relation R is equivalent to the ternary relation $\{(a, b, a) : (a, b) \in R\}$.

3.1 Operations on binary relations

A binary relation is a set of ordered pairs, denoted (a, b). For any two binary relations R and S, $R \cap S$ is the intersection of R and S, $R \cup S$ is the union of R and S, $R \circ S$ is the composition of R and S, and R^{\sim} is the converse of R; these are defined as follows:

$$\begin{array}{lll} R \cap S &=& \{(a,b): (a,b) \in R \text{ and } (a,b) \in S\}, \\ R \cup S &=& \{(a,b): (a,b) \in R \text{ or } (a,b) \in S\}, \\ R \circ S &=& \{(a,b): \text{for some } c, (a,c) \in R \text{ and } (c,b) \in S\}, \\ R^{\smile} &=& \{(a,b): (b,a) \in R\}. \end{array}$$

Three special binary relations over a universe U are the empty relation \emptyset which contains no pairs at all, the identity relation $\mathcal{I}_U^b = \{(a, a) : a \in U\}$, and the universal relation $\top_U^b = U \times U$.

3.2 Operations on ternary relations

A ternary relation is a set of ordered triples, denoted (a, b, c). For any two ternary relations R and S, $R \cap S$ is the intersection of R and S, $R \cup S$ is the union of R and S, $R \circ S$ is the composition of R and S, R^{\frown} is the converse of R, and R^{\frown} is the rotation of R; these are defined as follows:

$$\begin{array}{lll} R \cap S &=& \{(a,b,c):(a,b,c) \in R \text{ and } (a,b,c) \in S\}, \\ R \cup S &=& \{(a,b,c):(a,b,c) \in R \text{ or } (a,b,c) \in S\}, \\ R \circ S &=& \{(a,b,c): \text{for some } d, (a,b,d) \in R \text{ and } (a,d,c) \in S\}, \\ R^{\smile} &=& \{(a,b,c):(a,c,b) \in R\}, \\ R^{\frown} &=& \{(a,b,c):(c,a,b) \in R\}. \end{array}$$

In terms of expressiveness, it should be said that the converse and the rotation of a relation R record the same information as R itself. For binary relations, a converse operation is sufficient because there are two possible ordered pairs involving two objects, say x and y: (x, y) and (y, x); the converse operation alone allows going from one of the two pairs to the other. For ternary relations, a converse operation is no longer sufficient because there are altogether six possible ordered triples involving three objects, say x, y and z:

(x, y, z), (x, z, y), (y, x, z), (y, z, x), (z, x, y), (z, y, x). The converse operation allows going from an ordered triple (x, y, z) to the ordered triple (x, z, y), but does not allow going to the other four ordered triples. With the addition of the rotation operation, we can move as well to (y, z, x); then from (y, z, x) to (y, x, z) using converse, and to (z, x, y) using rotation; and from (x, z, y) to (z, y, x) using rotation.

Three special ternary relations over a universe U are the empty relation \emptyset which contains no triples at all, the identity relation $\mathcal{I}_U^t = \{(a, a, a) : a \in U\}$, and the universal relation $\top_U^t = U \times U \times U$. Another special ternary relation, which expresses equality of the last two arguments and will be needed later, is $\mathcal{I}_U^{t_{23}} = \{(a, b, b) : a, b \in U\}$.

The field of a binary relation R is $field(R) = \{a : \text{ for some } b, (a, b) \in R \text{ or } (b, a) \in R\}$; the field of a ternary relation R is $field(R) = \{a : \text{ for some } b \text{ and } c, (a, b, c) \in R \text{ or } (b, a, c) \in R \text{ or } (b, c, a) \in R\}$. The field of a set \mathcal{A} of relations is the union of the fields of the relations in \mathcal{A} : $field(\mathcal{A}) = \bigcup_{R \in \mathcal{A}} field(R)$.

3.3 Constraint matrices

Let P be a CSP of order n, with variables x_1, \ldots, x_n and universe U.

3.3.1 The case of a binary CSP

Let x_i, x_j be two variables. If a constraint of P is given on the ordered pair (x_j, x_i) , specifying that (x_j, x_i) should belong to a relation R, this can be converted into a constraint on the ordered pair (x_i, x_j) : $(x_i, x_j) \in R^{\smile}$. Therefore, we can assume that if m constraints involve the variables x_i and x_j then these constraints consist of binary relations R_1, \ldots, R_m the ordered pair (x_i, x_j) is required to belong to. These m constraints are then converted into the single constraint $(x_i, x_j) \in R_1 \cap \ldots \cap R_m$. We can therefore, without loss of generality, make the assumption that for any two variables x_i and x_j , there is at most one constraint involving x_i and x_j .

A binary constraint matrix of order n over U is an $n \times n$ -matrix of binary relations over U verifying the following:

$$\begin{array}{ll} (\forall i \leq n)(M_{ii} \subseteq \mathcal{I}_U^b) & (\text{the diagonal property}), \\ (\forall i, j \leq n)(M_{ij} = (M_{ji})^{\smile}) & (\text{the converse property}). \end{array}$$

A binary CSP P over a universe U can be associated with the following binary constraint matrix, denoted M^P :

- 1. Initialise all entries to the universal relation: $(\forall i, j \leq n)((M^P)_{ij} \leftarrow \top^b_U)$
- 2. Initialise the diagonal elements to the identity relation: $(\forall i \leq n)((M^P)_{ii} \leftarrow \mathcal{I}_U^b)$
- 3. For all pairs (x_i, x_j) of variables on which a constraint $(x_i, x_j) \in R$ is specified: $(M^P)_{ij} \leftarrow (M^P)_{ij} \cap R, (M^P)_{ji} \leftarrow ((M^P)_{ij})^{\smile}$.

3.3.2 The case of a ternary CSP

Let x_i, x_j, x_k be three variables; there are altogether six possible ordered triples on them: $(x_i, x_j, x_k), (x_i, x_k, x_j), (x_j, x_i, x_k), (x_j, x_k, x_i), (x_k, x_i, x_j), (x_k, x_j, x_i)$. If a constraint of P involving x_i, x_j and x_k is given on an ordered triple other than (x_i, x_j, x_k) , this can be converted into a constraint on the ordered triple (x_i, x_j, x_k) by using a finite combination of the converse and rotation operations. For instance, a constraint of the form $(x_k, x_j, x_i) \in R$ is equivalent to $(x_i, x_j, x_k) \in (R^{\frown})^{\frown}$. We can therefore assume that if m constraints of P involve the variables x_i, x_j, x_k then these consist of ternary relations R_1, \ldots, R_m the ordered triple (x_i, x_j, x_k) is required to belong to. These m constraints are then converted into the single constraint $(x_i, x_j, x_k) \in R_1 \cap \ldots \cap R_m$. We can therefore, without loss of generality, make the assumption that for any three variables x_i, x_j, x_k , there is at most one constraint involving them.

A ternary constraint matrix of order n over U is an $n \times n \times n$ -matrix of ternary relations over U verifying the following:

$(\forall i \leq n)(M_{iii} \subseteq \mathcal{I}_U^t)$	(the identity property),
$(\forall i, j, k \le n)(M_{ijk} = (M_{ikj})^{\smile})$	(the converse property),
$(\forall i, j, k \leq n) (M_{ijk} = (M_{kij})^{\frown})$	(the rotation property).

A ternary CSP P over a universe U can be associated with the following ternary constraint matrix, denoted M^P :

- 1. Initialise all entries to the universal relation: $(\forall i, j, k \leq n)((M^P)_{ijk} \leftarrow \top^t_U)$
- 2. Initialise the diagonal elements to the identity relation: $(\forall i \leq n)((M^P)_{iii} \leftarrow \mathcal{I}_U^t)$
- 3. For all triples (x_i, x_j, x_k) of variables on which a constraint $(x_i, x_j, x_k) \in R$ is specified: $(M^P)_{ijk} \leftarrow (M^P)_{ijk} \cap R, \quad (M^P)_{ikj} \leftarrow ((M^P)_{ijk})^{\frown}, (M^P)_{jki} \leftarrow ((M^P)_{ijk})^{\frown}, (M^P)_{jki} \leftarrow ((M^P)_{jki})^{\frown}, (M^P)_{kjj} \leftarrow ((M^P)_{jki})^{\frown}, (M^P)_{kjj} \leftarrow ((M^P)_{kjj})^{\frown}.$

We make the assumption that, unless explicitly specified otherwise, a CSP is given as a constraint matrix.

3.4 Strong *k*-consistency, refinement

Let P be a CSP of order n, V its set of variables and U its universe. An instantiation of P is any n-tuple (a_1, a_2, \ldots, a_n) of U^n , representing an assignment of a value to each variable. A consistent instantiation is an instantiation (a_1, a_2, \ldots, a_n) which is a solution:

• If P is a binary CSP: $(\forall i, j \leq n) ((a_i, a_j) \in (M^P)_{ij});$

• If P is a ternary CSP: $(\forall i, j, k \leq n)((a_i, a_j, a_k) \in (M^P)_{ijk}).$

P is consistent if it has at least one solution; it is inconsistent otherwise. The consistency problem of P is the problem of verifying whether P is consistent.

Let $V' = \{x_{i_1}, \ldots, x_{i_j}\}$ be a subset of V. The sub-CSP of P generated by V', denoted $P_{|V'}$, is the CSP with set of variables V' and whose constraint matrix is obtained by projecting the constraint matrix of P onto V':

- If P is a binary CSP then: $(\forall k, l \leq j)((M^{P_{|V'}})_{kl} = (M^P)_{i_k i_l})$
- If P is a ternary CSP then: $(\forall k, l, m \leq j)((M^{P_{|V'|}})_{klm} = (M^P)_{i_k i_l i_m})$

P is *k*-consistent [13, 14] if for any subset *V'* of *V* containing k-1 variables, and for any variable $X \in V$, every solution to $P_{|V'|}$ can be extended to a solution to $P_{|V'| \in X^3}$. *P* is strongly *k*-consistent if it is *j*-consistent, for all $j \leq k$.

1-consistency, 2-consistency and 3-consistency correspond to node-consistency, arc-consistency and path-consistency, respectively [26, 28]. Strong *n*-consistency of P corresponds to what is called global consistency in [6]. Global consistency facilitates the important task of searching for a solution, which can be done, when the property is met, without backtracking [14].

A refinement of P is a CSP P' with the same set of variables and such that

- $(\forall i, j)((M^{P'})_{ij} \subseteq (M^P)_{ij})$, in the case of binary CSPs.
- $(\forall i, j, k)((M^{P'})_{ijk} \subseteq (M^{P})_{ijk})$, in the case of ternay CSPs.

4 Relation algebras

We will be using unary operators $(-, \smile \text{ and } \frown)$ and binary operators $(\oplus, \odot \text{ and } \circ)$. In expressions without full parentheses, unary operators should be computed first, followed by \circ , \odot , and \oplus , in that order.

A Boolean algebra with universe \mathcal{B} is an algebra of the form $\langle \mathcal{B}, \oplus, \odot, \bar{}, \bot, \top \rangle$ which satisfies the following properties, for all $R, S, T \in \mathcal{B}$:

$$R \oplus (S \oplus T) = (R \oplus S) \oplus T,$$

$$R \oplus S = S \oplus R,$$

$$R \odot S \oplus R = R,$$

$$R \odot S \oplus T = (R \oplus T) \odot (S \oplus T)$$

$$R \oplus \overline{R} = \top.$$

Of particular interest to this work are Boolean algebras of the form $\langle 2^A, \cup, \cap, -, \emptyset, A \rangle$, where A is a nonempty finite set.

We will consider exclusively relation algebras (henceforth, RAs) of which the universe is a set of binary relations, and RAs of which the universe is a set of ternary relations; we refer to the former as binary RAs, to the latter as ternary RAs.

4.1 Binary RAs

 \mathcal{U} is a binary RA with universe \mathcal{A} if:

- 1. \mathcal{A} is a set of binary relations; and
- U = ⟨A, ⊕, ⊙, ¬, ⊥, ⊤, ∘, ¬, ⊥) where ⟨A, ⊕, ⊙, ¬, ⊥, ⊤⟩ is a Boolean algebra (called the Boolean part, or reduct, of U), ∘ is a binary operation, ~ is a unary operation, I ∈ A, and the following identities hold for all R, S, T ∈ A:

 $\begin{array}{l} (R \circ S) \circ T = R \circ (S \circ T), \\ (R \oplus S) \circ T = R \circ T \oplus S \circ T, \\ R \circ \mathcal{I} = \mathcal{I} \circ R = R, \\ (R^{\frown})^{\frown} = R, \\ (R \oplus S)^{\frown} = R^{\frown} \oplus S^{\frown}, \\ (R \circ S)^{\frown} = S^{\frown} \circ R^{\frown}, \\ R^{\frown} \circ \overline{R \circ S} \odot S = \bot. \end{array}$

The properties that hold for a binary RA can be seen as the minimal properties that hold for an RA whose universe is a set of *m*-ary relations, with $m \ge 2$. These correspond to the list of properties of RAs given by Tarski [35]. When the universe is a set of *m*-ary relations, with $m \ge 3$, further properties arise, due to further operations. Tarski provided for RAs the above list of properties because he was interested mainly in formalising the theory of binary relations.

We now describe ternary RAs, which need an additional operation (rotation) and therefore additional properties.

4.2 Ternary RAs

 \mathcal{U} is a ternary RA with universe \mathcal{A} if:

- 1. \mathcal{A} is a set of ternary relations; and
- 2. $\mathcal{U} = \langle \mathcal{A}, \oplus, \odot, \bar{}, \bot, \top, \circ, \bar{}, \mathcal{I} \rangle$ where $\langle \mathcal{A}, \oplus, \odot, \bar{}, \bot, \top \rangle$ is a Boolean algebra (called the Boolean part, or reduct, of \mathcal{U}), \circ is a binary operation, $\tilde{}$ and $\hat{}$ are unary operations, $\mathcal{I} \in \mathcal{A}$, and the following identities hold for all $R, S, T \in \mathcal{A}$:

$$(R \circ S) \circ T = R \circ (S \circ T),$$

$$(R \oplus S) \circ T = R \circ T \oplus S \circ T,$$

$$R \circ \mathcal{I} = \mathcal{I} \circ R = R,$$

$$(R^{\frown})^{\frown} = R,$$

$$(R \oplus S)^{\frown} = R^{\frown} \oplus S^{\frown},$$

$$(R \circ S)^{\frown} = S^{\frown} \circ R^{\frown},$$

$$R^{\frown} \circ \overline{R \circ S} \odot S = \bot,$$

$$(R \oplus S)^{\frown} = R^{\frown} \oplus S^{\frown}.$$

4.3 Atomic RA

An atom of an RA \mathcal{U} is a minimal nonzero element, i.e., R is an atom if $R \neq \bot$ and for every $S \in \mathcal{A}$, either $R \odot S = \bot$ or $R \odot \overline{S} = \bot$. An RA is atomic if every nonzero element has an atom below it; i.e., if for all nonzero elements R, there exists an atom A such that $A \odot R = A$.

In the rest of the paper, we focus on atomic, finite RAs of which the Boolean part is of the form $\langle 2^{\top}, \cup, \cap, \bar{}, \emptyset, \top \rangle$:

- 1. The top element \top is a finite set of atoms; the bottom element \bot is the empty set \emptyset ; the universe is the set 2^{\top} of all subsets of \top ; and
- 2. the operations \oplus , \odot and $^-$ are the usual set-theoretic operations of union (\cup) , intersection (\cap) and complement $(^-)$ with respect to \top (i.e., $(\forall R \in 2^{\top})(\overline{R} = \top \setminus R)$).

A finite RA is atomic, and its Boolean part is completely determined by its atoms. Furthermore, in an atomic RA, the result of applying any of the operations of the RA to any elements can be obtained from the results of applying the different operations to the atoms. Specifying a finite, thus atomic, RA reduces thus to specifying the identity element and the results of applying the different operations to the different atoms.

5 An atomic binary RA of 2D orientations

We introduce in this section the first RA of 2D orientations.

5.1 The field

The field $field(\mathcal{U})$ of an RA \mathcal{U} with universe \mathcal{A} is the union of the fields of the relations in \mathcal{A} ; i.e., $field(\mathcal{U}) = \bigcup_{R \in \mathcal{A}} field(R)$. The field of the RA to be introduced is the set 2D \mathcal{O} of 2D orientations.

5.2 The universe

Given an orientation X of the plane, another orientation Y can form with X one of the following qualitative configurations:

- 1. Y is equal to X: the angle (X, Y) is equal to 0.
- 2. Y is to the left of X: the angle (X, Y) belongs to $(0, \pi)$.
- 3. Y is opposite to X: the angle (X, Y) is equal to π .
- 4. Y is to the right of X: the angle (X, Y) belongs to $(\pi, 2\pi)$.
 - 13

b	e	l	0	r
b^{\smile}	е	r	0	l

0	e	l	0	r
e	e	l	0	r
l	l	$\{l, o, r\}$	r	$\{e, l, r\}$
0	0	r	e	l
r	r	$\{e, l, r\}$	l	$\{l, o, r\}$

Figure 3: (Left) The converse b^{\smile} of a \mathcal{CYC}_b atom b; (Right) The composition for every pair of \mathcal{CYC}_b atoms: the entry on row *i*, column *j* is the \mathcal{CYC}_b relation consisting of the composition of the leftmost element of the row and the top element of the column.

We denote the four configurations by e(Y,X), l(Y,X), o(Y,X) and r(Y,X), respectively. The configurations are Jointly Exhaustive and Pairwise Disjoint (JEPD): given any two orientations of the plane, they stand in one and only one of the configurations.

Definition 3 (the atoms) The RA contains four atoms: e, l, o, r. We will refer to the set of all atoms as BIN.

BIN is the universal binary relation over 2D \mathcal{O} : BIN $\equiv \top_{2D\mathcal{O}}^{b} \equiv 2D\mathcal{O} \times 2D\mathcal{O}$.

Definition 4 (the universe) The universe of the RA, i.e., the set of all its relations, is the set of subsets of BIN. An element B of the universe is to be interpreted as follows: $(\forall X, Y \in 2D\mathcal{O})(B(Y, X) \Leftrightarrow \bigvee_{b \in B} b(Y, X))$

We refer to the set of singleton relations as \mathcal{AT}_b : $\mathcal{AT}_b = \{\{e\}, \{l\}, \{o\}, \{r\}\}\}$. We notice that \mathcal{AT}_b is a set of relations, whereas *BIN* is a relation. When there is no risk of confusion, we omit the braces in the representation of a singleton relation.

5.3 The operations applied to the atoms

Figure 3(Left) gives the converse for each of the atoms. Figure 3(Right) gives the composition for every pair of atoms.

5.4 The identity element

The identity element is the atom e; the composition table of Figure 3(Right) can be used to verify that: $(\forall R \in 2^{BIN})(R \circ e = e \circ R = R)$.

The RA so defined is an atomic binary RA, which we name \mathcal{CYC}_b : $\mathcal{CYC}_b = \langle 2^{BIN}, \cup, \cap, -, \emptyset, BIN, \smile, e \rangle$. BIN is the universal \mathcal{CYC}_b relation: $(\forall X, Y \in 2D\mathcal{O})(BIN(Y, X))$.

The structure of \mathcal{CYC}_b is very similar to Allen's algebra of temporal intervals [1], presented by Ladkin and Maddux as an atomic binary RA [20]. In Appendix A, we verify the RA properties for \mathcal{CYC}_b .

5.5 Additional definitions

We make use of the isomorphism $f_1^{2D\mathcal{O}\rightarrow}$ alluded to earlier, from the set $2D\mathcal{O}$ onto the set of radii of circle $\mathcal{C}_{O,1}$: as we have seen, given an orientation z, $f_1^{2D\mathcal{O}\rightarrow}(z)$ is the radius $(OP_z]$ of $\mathcal{C}_{O,1}$, excluding the centre O, such that the orientation of the vector $\overrightarrow{OP_z}$ is z.

Definition 5 (sector of a $C\mathcal{YC}_b$ **relation)** The sector determined by an orientation z and a $C\mathcal{YC}_b$ relation B, denoted sect(z, B), is the sector of circle $C_{O,1}$, excluding the centre O, representing the set of orientations z' related to z by the relation B: $sect(z, B) = \{f_1^{2DO \rightarrow}(z') | B(z', z)\}.$

Remark 1 The sector determined by an orientation and a $C\mathcal{YC}_b$ relation does not include the centre O of circle $C_{O,1}$. Therefore, given n orientations z_1, \ldots, z_n

and $n \ C \mathcal{YC}_b$ relations B_1, \ldots, B_n , the intersection $\bigcap_{i=1}^n sect(z_i, B_i)$ is either the

empty set or a set of radii:¹ this cannot be equal to the centre O, which would be possible if the sector determined by an orientation and a CYC_b relation included O. This is important for the understanding of the proof of Theorem 6.

Definition 6 Let B be a CYC_b relation:

- 1. B is convex if for all orientations z, sect(z, B) is a convex part of the plane.
- 2. The dimension of B is the dimension of the sector it determines with any orientation.
- 3. B is holed if:
 - (a) it is equal to BIN; or
 - (b) the difference $BIN \setminus B$ is a CYC_b relation of dimension 1 (is equal to e, o or $\{e, o\}$).

The two atoms e and o of CYC_b are of dimension 1, the other two (l and r) of dimension 2. Moreover, the dimension of a CYC_b relation in general is the greatest of the dimensions of its atoms.

 $^{^1\}mathrm{A}$ set of radii represents, according to our convention (Definition 1), a set of orientation values.



Figure 4: The ternary relation induced from three $C\mathcal{YC}_b$ atoms: $b_1b_2b_3(X, Y, Z)$ iff $b_1(Y, X) \wedge b_2(Z, Y) \wedge b_3(Z, X)$.

Intuitively, a $C \mathcal{YC}_b$ relation is holed if the sector it determines with any orientation is almost equal to the entire surface of circle $C_{O,1}$; i.e., the topological closure of the sector is equal to the entire surface.

We will refer to the set of all \mathcal{CYC}_b relations which are either convex or holed as BCH. BCH splits into:

- 1. eight convex relations: $\{e\}, \{l\}, \{o\}, \{r\}, \{e, l\}, \{e, r\}, \{l, o\}, \{o, r\}$; and
- 2. four holed relations: $\{l, r\}, \{e, l, r\}, \{l, o, r\}, \{e, l, o, r\}$.

Notice that neither of the $C\mathcal{YC}_b$ relations $\{e, l, o\}$ and $\{e, o, r\}$ is convex. For instance, the sector determined by an orientation, say z, and the former relation, $\{e, l, o\}$, is equal to π minus the centre of $C_{0,1}$.

Definition 7 (induced ternary relation) Given three CYC_b atoms b_1, b_2, b_3 , we define the induced ternary relation $b_1b_2b_3$ as follows (see Figure 4):

$$(\forall X, Y, Z)(b_1b_2b_3(X, Y, Z) \Leftrightarrow b_1(Y, X) \land b_2(Z, Y) \land b_3(Z, X))$$

The \mathcal{CYC}_b composition table (Figure 3(Right)) has 12 entries consisting of atoms, the remaining four consisting of three-atom relations. Therefore any three 2D orientations stand in one of the following 24 JEPD configurations: eee, ell, eoo, err, lel, lll, llo, llr, lor, lre, lrl, lrr, oeo, olr, ooe, orl, rer, rle, rll, rlr, rol, rrl, rro, rrr. According to Definition 7, rol(X, Y, Z), for instance, means $r(Y, X) \wedge o(Z, Y) \wedge l(Z, X)$.

The composition table rules out the other, $(4 \times 4 \times 4) - 24$, induced ternary relations $b_1b_2b_3$; these are inconsistent: no triple (z_1, z_2, z_3) of orientations exists such that for such an induced relation one has $b_1(z_2, z_1) \wedge b_2(z_3, z_2) \wedge b_3(z_3, z_1)$.

6 An atomic ternary RA of 2D orientations

As we will see, the CYCORD relation cyc [31, 32] holds on a triple (z_1, z_2, z_3) of 2D orientations if the images P_{z_1} , P_{z_2} and P_{z_3} of z_1 , z_2 and z_3 , respectively, of the isomorphism $f_2^{2D\mathcal{O}}$ (Definition 1) are:

- 1. pairwise distinct, and
- 2. such that P_{z_2} is met before P_{z_3} when we scan the circle $\mathcal{C}_{O,1}$ in a clockwise direction starting from P_{z_1} .

The RA $C \mathcal{Y} \mathcal{C}_b$ cannot represent the relation *cyc*. However, we can define an atomic ternary RA of which the atoms are the "*induced ternary relations*" described above, which will have *cyc* as one of the elements of its universe.

6.1 The field

As for $C \mathcal{YC}_b$, the field of this new RA, which we name $C \mathcal{YC}_t$, is the set 2D \mathcal{O} of 2D orientations.

6.2 The universe

Definition 8 (the atoms) An atom of $C\mathcal{YC}_t$ is any of the 24 JEPD configurations a triple of 2D orientations can stand in. We denote the set of all atoms by TER: $TER = \{eee, ell, eoo, err, lel, lll, llo, llr, lor, lre, lrl, lrr, oeo, olr, ooe,$ $orl, rer, rle, rll, rlr, rol, rrl, rro, rrr\}.$

TER is the universal ternary relation over 2D \mathcal{O} : *TER* $\equiv \top_{2D\mathcal{O}}^t \equiv 2D\mathcal{O} \times 2D\mathcal{O} \times 2D\mathcal{O}$.

Definition 9 (The universe) The universe of the ternary RA, i.e., the set of all its relations, is the set of subsets of TER. An element T of the universe is to be interpreted as follows: $(\forall X, Y, Z \in 2DO)(T(X, Y, Z) \Leftrightarrow \bigvee_{t \in T} t(X, Y, Z))$

We refer to the set of singleton relations as \mathcal{AT}_t : $\mathcal{AT}_t = \{\{t\} : t \in TER\}$. When there is no risk of confusion, we omit the braces in the representation of a singleton relation.

6.3 The operations applied to the atoms

Figure 5 gives the converse and the rotation for each of the 24 atoms.

In order to give a simple way of writing the composition tables, we have to look closely at how composition is computed. Given four 2D orientations X, Y, Z, W and two atoms t_1 and t_2 , corresponding, respectively, to the induced ternary relations $b_1b_2b_3$ and $b'_1b'_2b'_3$, the conjunction $t_1(X, Y, Z) \wedge t_2(X, Z, W)$ is inconsistent if $b_3 \neq b'_1$ (see Figure 6 for illustration); this is because the CYC_b

t	t	t^{\frown}	t	t	t^{\frown}	t	t	t^{\frown}		t	t	t^{\frown}
eee	eee	eee	llo	orl	lor	oeo	oeo	eoo]	rll	lrr	lrl
ell	lre	lre	llr	rrl	llr	olr	rro	llo		rlr	rrr	lll
eoo	<i>00e</i>	<i>00e</i>	lor	rol	olr	<i>00e</i>	eoo	oeo		rol	lor	orl
err	rle	rle	lre	ell	rer	orl	llo	rro		rrl	llr	rrl
lel	lel	err	lrl	lll	rrr	rer	rer	ell		rro	olr	rol
111	lrl	lrr	lrr	rll	rlr	rle	err	lel		rrr	rlr	rll

Figure 5: The converse t^{\frown} and the rotation t^{\frown} of a CYC_t atom t.



Figure 6: The conjunction $b_1b_2b_3(X,Y,Z)\wedge b_1'b_2'b_3'(X,Z,W)$ is inconsistent if $b_3\neq b_1'.$

atoms are JEPD. Stated otherwise, when $b_3 \neq b'_1$ we have $t_1 \circ t_2 = \emptyset$. Thus composition splits into four composition tables, corresponding to the following four cases:

- 1. Case 1: $b_3 = b'_1 = e$. This corresponds to $t_1 \in \{eee, lre, ooe, rle\}$ and $t_2 \in \{eee, ell, eoo, err\}$.
- 2. Case 2: $b_3 = b'_1 = l$. This corresponds to $t_1 \in \{ell, lel, lll, lrl, orl, rll, rol, rrl\}$ and $t_2 \in \{lel, lll, llo, llr, lor, lre, lrl, lrr\}$.
- 3. Case 3: $b_3 = b'_1 = o$. This corresponds to $t_1 \in \{eoo, llo, oeo, rro\}$ and $t_2 \in \{oeo, olr, ooe, orl\}$.
- 4. Case 4: $b_3 = b'_1 = r$. This corresponds to $t_1 \in \{err, llr, lor, lrr, olr, rer, rlr, rrr\}$ and $t_2 \in \{rer, rle, rll, rlr, rol, rrl, rro, rrr\}$.

Figure 7 presents the four composition tables.²

6.4 The identity element

Given a universe U, we have defined the relation $\mathcal{I}_U^{t_{23}}$ as $\{(a, b, b) : a, b \in U\}$. $\mathcal{I}_U^{t_{23}}$ expresses equality of the last two arguments, and leaves unspecified the relation between the first two. Since $b_1b_2b_3(X,Y,Z)$, where $b_1b_2b_3$ is a \mathcal{CYC}_t atom, represents the conjunction $b_1(Y,X) \wedge b_2(Z,Y) \wedge b_3(Z,X)$, this means that if Y and Z are equal then for $b_1b_2b_3(X,Y,Z)$ to hold, b_2 must be e, and b_1 and b_3 must be identical. Thus, when U coincides with 2D \mathcal{O} , we get:

$$\mathcal{I}_{2\mathrm{D}\mathcal{O}}^{t_{23}} = \{eee, lel, oeo, rer\}$$

Using the composition tables, we can verify that $\mathcal{I}_{2D\mathcal{O}}^{t_{23}}$ is an identity element for \mathcal{CYC}_t :

$$(\forall R \in 2^{TER})(R \circ \mathcal{I}_{2D\mathcal{O}}^{t_{23}} = \mathcal{I}_{2D\mathcal{O}}^{t_{23}} \circ R = R)$$

This completes the presentation of \mathcal{CYC}_t :

$$\mathcal{CYC}_{t} = \langle 2^{TER}, \cup, \cap, \bar{}, \emptyset, TER, \bar{}, \uparrow, \mathcal{I}_{2D\mathcal{O}}^{t_{23}} \rangle$$

TER is the universal $C\mathcal{YC}_t$ relation: $(\forall X, Y, Z \in 2D\mathcal{O})(TER(X, Y, Z))$.

In Appendix B, we verify the RA properties for an atomic ternary RA.

²Alternatively, one could define one single composition table for \mathcal{CYC}_t . Such a table would have 24×24 entries, most of which (i.e., $24 \times 24 - (16 + 64 + 16 + 64)$) would be the empty relation.

			0	eee	ell	<i>eoo</i>	err			
			eee	eee	ell	<i>eoo</i>	err			
			lre	lre	$\{lel, lll, lrl\}$	llo	$\{llr,$	$lor, lrr\}$		
			00e	00e	orl	0e0	olr			
			rle	rle	$\{rll, rol, rrl\}$	rro	$\{rer\}$	[rlr, rrr]	}	
0	lel	111		llo	llr	lor	lre	lrl		lrr
ell	ell	ell		eoo	err	err	eee	ell		err
lel	lel	111		llo	llr	lor	lre	lrl		lrr
111	lll	Ш		llo	$\{llr, lor, lrr\}$	lrr	lre	$\{lel, lll,$	lrl	lrr
lrl	lrl	$\{lel,l\}$	ll,lrl}	llo	llr	llr	lre	lrl		$\{llr, lor, lrr\}$
orl	orl	orl		oeo	olr	olr	00e	orl		olr
rll	rll	${rll,r}$	ol,rrl}	rro	rrr	rrr	rle	rll		$\{rer, rlr, rrr\}$
rol	rol	rrl		rro	rrr	rer	rle	rll		rlr
rrl	rrl	rrl		rro	$\{rer, rlr, rrr\}$	rlr	rle	${rll.rol}$.rrl	rlr
-								ι /	/]	
	• <u>·</u> ····		0	0e0	olr	00e	orl	ι ()	<u>, א</u>	
			0 eoo	0e0 e00	olr err	00e eee	orl ell		 	
<u> </u>			○ <i>eoo</i> <i>llo</i>	oeo eoo llo	olr err {llr, lor, lrr}	ooe eee lre	orl ell {lel	, lll, lrl}		
			 ○ eoo llo oeo 	0e0 e00 llo 0e0	olr err {llr, lor, lrr} olr	ooe eee lre ooe	orl ell {lel orl	, <i>lll</i> , <i>lrl</i> }		
			○ eoo llo oeo rro	0e0 e00 ll0 0e0 rr0	olr err {llr, lor, lrr} olr {rer, rlr, rrr}	ooe eee lre ooe rle	orl ell {lel orl {rll	, lll, lrl} , rol, rrl]	 }	
0	rer	rle	 ○ eoo llo oeo rro 	0e0 e00 ll0 0e0 rr0	olr err {llr, lor, lrr} olr {rer, rlr, rrr} rlr	00e eee lre 00e rle rol	orl ell {lel orl {rll rrl	, lll, lrl} , rol, rrl]	rro	rrr
o err	rer err	rle eee	 ○ eoo llo oeo rro rll ell 	0e0 e00 ll0 0e0 rr0	olr err {llr, lor, lrr} olr {rer, rlr, rrr} rlr err	ooeeeelreooerlerolell	orl ell {lel orl {rll ell	, lll, lrl} , rol, rrl]	rro eoo	rrr err
∘ err llr	rer err llr	rle eee lre	 ○ eoo llo oeo rro rll ell lrl 	0e0 e00 ll0 0e0 rr0	olr err {llr, lor, lrr} olr {rer, rlr, rrr} rlr err {llr, lor, lrr}	00e eee lre 00e rle rol ell lrl	orl ell {lel orl {rll rrl ell {lel,	, lll, lrl} , rol, rrl] ,lrl}	<pre> rro eoo llo </pre>	rrr err llr
o err llr lor	rer err llr lor	rle eee lre lre	• eoo llo oeo rro rll ell lrl lrl	0e0 e00 ll0 0e0 rr0	olr err {llr, lor, lrr} olr {rer, rlr, rrr} rlr err {llr, lor, lrr} lrr	00e eee lre 00e rle rol ell lrl lel	orl ell {lel orl {rll rrl ell {lel,	, lll, lrl} , rol, rrl] lll,lrl}	rro eoo llo llo	rrr err llr llr
∘ err llr lor lrr	rer err llr lor lrr	rle eee lre lre lre	• eoo llo oeo rro rll ell lrl lrl {lel,l	oeo eoo llo oeo rro	olr err {llr, lor, lrr} olr {rer, rlr, rrr} rlr err {llr, lor, lrr} lrr lrr	00e eee lre 00e rle rol ell lrl lel lll	orl ell {lel orl {rll rrl ell {lel, lll lll	, lll, lrl} , rol, rrl] 	rro eoo llo llo llo	rrr err llr llr {llr,lor,lrr}
o err llr lor lrr olr	rer err llr lor lrr olr	rle eee lre lre lre ooe	• eoo llo oeo rro rll ell lrl lrl lell, lor orl	oeo eoo llo oeo rro ll,lrl}	olr err {llr, lor, lrr} olr {rer, rlr, rrr} rlr err {llr, lor, lrr} lrr olr olr	00e eee lre 00e rle rol ell lrl lel orl	orl ell {lel orl {rll rrl ell {lel, lll lll orl	, lll, lrl} , rol, rrl] 	rro eoo llo llo llo oeo	rrr err llr llr {llr,lor,lrr} olr
o err llr lor lrr olr rer	rer err llr lor lrr olr rer	rle eee lre lre lre ooe rle	• eoo llo oeo rro rll ell lrl lrl orl rll	oeo eoo llo oeo rro ll,lrl}	olr err {llr, lor, lrr} olr {rer, rlr, rrr} rlr err {llr, lor, lrr} lrr olr rlr	00e eee lre 00e rle rol ell lrl lel lll orl rol	orl ell {lel orl {rll ell {lel, lll orl rrl	, lll, lrl} , rol, rrl] lll,lrl}	rro eoo llo llo llo oeo rro	rrr err llr llr {llr,lor,lrr} olr rrr
o err llr lor lrr olr rer rlr	rer err llr lor lrr olr rer rlr	rle eee lre lre lre ooe rle rle	• eoo llo oeo rro rll ell lrl lrl lel, lo orl rll fll, le, l orl rll	oeo eoo llo oeo rro	olr olr err {llr, lor, lrr} olr {rer, rlr, rrr} rlr err {llr, lor, lrr} lrr olr rlr rr rr	000 eee lre 000 rle rol lrl lel lll orl rol	orl ell {lel orl {rll ell {lel, lll lll orl rrl	, lll, lrl} , rol, rrl] ill, lrl}	rro eoo llo llo llo rro rro rro	rrr err llr llr {llr,lor,lrr} olr rrr {rer,rlr,rrr}

Figure 7: The \mathcal{CYC}_t composition tables: case 1, case 2, case 3 and case 4, respectively, from top to bottom.



Figure 8: Graphical illustration of the 24 CYC_t atoms: from top to bottom, left to right, the atoms are lrl, orl, rll, rol, rrl, rro, rrr, lll, llo, lrr, lor, llr, olr, rlr, eee, ell, eoo, err, lel, oeo, rer, lre, ooe, rle.

6.5 Examples

Example 1 For each CYC_t atom t, Figure 8 presents a configuration of orientations X, Y and Z such that t(X, Y, Z) holds:

- The top row illustrates, from left to right, the atoms lrl, orl, rll, rol, rrl, rro, rrr.
- The second row from the top illustrates, from left to right, the atoms lll, llo, lrr, lor, llr, olr, rlr.
- The third row from the top illustrates, from left to right, the atoms eee, ell, eoo, err, lel, oeo, rer.
- Finally, the bottom row illustrates, from left to right, the atoms lre, ooe, rle.

Example 2 Consider again Figure 8:

- 1. Each atom illustrated on the second and fourth rows from the top is the converse of the atom illustrated just above it, on the preceding row.
- 2. The first and last three illustrations of the third row from the top have nothing underneath them, on the bottom row: each of the corresponding atoms is its proper converse.
- 3. Consider the relation cyc defined on the set 2DO as follows:

 $(\forall X,Y,Z \in 2 \mathrm{D}\,\mathcal{O})(\mathit{cyc}(X,\,Y,\,Z) \Leftrightarrow \, Y \neq X \land Z \neq \, Y \land \, Z \neq X \land \mathit{cw}\,(X,\,Y,Z))$

The relation cw holds on a triple (X, Y, Z) of 2D orientations if and only if we first meet Y and then Z when we move in a clockwise direction starting from X. Thus the relation cyc expresses strict betweenness in a clockwise direction. This relation is the unique relation of the CYCORD theory [31, 32], and is indeed an element of $C\mathcal{YC}_t$: cyc = {lrl, orl, rll, rol, rrl, rro, rrr} (the set of all atoms illustrated on the top row).

4. The converse of cyc is the set of all atoms illustrated on the second row from the top:

 $cyc = \{ lll, llo, lrr, lor, llr, olr, rlr \}$

Example 3 The composition rule for the CYCORD theory is as follows, and can be verified using the CYC_t composition tables:

 $(\forall X, Y, Z, W)(\mathit{cyc}(X, \, Y, \, Z) \land \mathit{cyc}(X, \, Z, W) \Rightarrow \mathit{cyc}(X, \, Y, \, W))$

²²

6.6 Additional definitions

Definition 10 (cross product of $C\mathcal{YC}_b$ **relations)** The cross product of three $C\mathcal{YC}_b$ relations B_1, B_2, B_3 , denoted $\Pi(B_1, B_2, B_3)$, is the $C\mathcal{YC}_t$ relation consisting of those atoms $b_1b_2b_3$ such that $b_1 \in B_1, b_2 \in B_2, b_3 \in B_3$:

 $\Pi(B_1, B_2, B_3) = \{b_1 b_2 b_3 : b_1 \in B_1, b_2 \in B_2, b_3 \in B_3\} \cap TER$

Definition 11 Let R be a CYC_t relation:

- 1. The first, second and third projections of R are the \mathcal{CYC}_b relations $\nabla^1(R)$, $\nabla^2(R)$ and $\nabla^3(R)$, respectively, defined as follows: $\nabla^1(R) = \{b_1 \in BIN : (\exists b_2, b_3 \in BIN)(b_1b_2b_3 \in R)\},$ $\nabla^2(R) = \{b_2 \in BIN : (\exists b_1, b_3 \in BIN)(b_1b_2b_3 \in R)\},$ $\nabla^3(R) = \{b_3 \in BIN : (\exists b_1, b_2 \in BIN)(b_1b_2b_3 \in R)\}$
- 2. R is projectable if $R = \Pi(\nabla^1(R), \nabla^2(R), \nabla^3(R))$.
- 3. R is convex if it is projectable, and each of its projections is a convex CYC_b relation.
- 4. R is said to be {convex,holed} (convex or holed) if it is projectable, and each of its projections is a CYC_b relation which is either convex or holed (belongs to BCH).

We notice that, given a \mathcal{CYC}_t relation R, $\bigtriangledown^1(R)$, $\bigtriangledown^2(R)$ and $\bigtriangledown^3(R)$ are the most specific \mathcal{CYC}_b relations such that:

$$(\forall X, Y, Z)(R(X, Y, Z) \Rightarrow \bigtriangledown^1(R)(Y, X) \land \bigtriangledown^2(R)(Z, Y) \land \bigtriangledown^3(R)(Z, X))$$

Example 4 1. $\Pi(\{e, o\}, \{l\}, \{l, r\}) = \{ell, olr\}.$

- 2. Let $R = \{ell, llo\}$. We have the following: $\nabla^1(R) = \{e, l\}, \nabla^2(R) = \{l\}$ and $\nabla^3(R) = \{l, o\}$.
- 3. The cross product of the three projections of the relation R above is $\Pi(\bigtriangledown^{1}(R), \bigtriangledown^{2}(R), \bigtriangledown^{3}(R)) = \Pi(\{e, l\}, \{l\}, \{l, o\}) = \{ell, lll, llo\}. \text{ Thus } R \neq \Pi(\bigtriangledown^{1}(R), \bigtriangledown^{2}(R), \bigtriangledown^{3}(R)), \text{ and } R \text{ is not projectable.}$

The set of all projectable CYC_t relations can be enumerated by computing for every three CYC_b relations their cross product. The set contains 1518 elements, including the empty relation.

We will refer to the subset of all {convex,holed} $C \mathcal{YC}_t$ relations as TCH.

Definition 12 (closures) Let S denote a subset of CYC_t . The weak closure of S is the smallest subset S^{wc} of CYC_t verifying the following properties:

(P1) $\mathcal{S} \subseteq \mathcal{S}^{wc}$; and

 $(P2) \ (\forall R, S \in \mathcal{S}^{wc})(R^{\smile} \in \mathcal{S}^{wc}, R^{\frown} \in \mathcal{S}^{wc}, R \cap S \in \mathcal{S}^{wc}).$

The closure of S under strong 4-consistency, or s4c-closure of S, is the smallest subset S^{s4c} of CYC_t verifying the following properties:

(P3) $\mathcal{S} \subseteq \mathcal{S}^{s4c}$; and

 $(P4) \ (\forall R, S, T \in \mathcal{S}^{s4c}) (R^{\smile} \in \mathcal{S}^{s4c}, R^{\frown} \in \mathcal{S}^{s4c}, R \cap S \in \mathcal{S}^{s4c}, R \circ S \cap T \in \mathcal{S}^{s4c}).$

The closure of S is the smallest subset S^c of CYC_t verifying the following properties:

- (P5) $\mathcal{S} \subseteq \mathcal{S}^c$; and
- $(P6) \ (\forall R, S \in \mathcal{S}^c)(R^{\smile} \in \mathcal{S}^c, R^{\frown} \in \mathcal{S}^c, R \cap S \in \mathcal{S}^c, R \circ S \in \mathcal{S}^c).$

Given a subset S of \mathcal{CYC}_t , we have $S^{wc} \subseteq S^{s4c} \subseteq S^c$. The relations in S^c can be viewed as resulting from the "execution" of (well-formed) expressions constructed from the alphabet $V_S = S \cup \{ \smile, \frown, \cap, \circ, (,) \}$; we refer to such expressions as S^c -expressions, and to the set of all of them as $Xp(S^c)$.

Definition 13 $Xp(S^c)$ is the smallest set of expressions over V_S verifying the following two properties:

- 1. a $C \mathcal{YC}_t$ relation belonging to S belongs to $Xp(S^c)$; and
- 2. if e_1 and e_2 belong to $Xp(\mathcal{S}^c)$ then so do $(e_1)^{\frown}, (e_1)^{\frown}, e_1 \cap e_2, e_1 \circ e_2$.

We suppose the reader familiar with (labelled) binary trees (each node of such a tree has at most two immediate successors). If a (binary) tree t reduces to a leaf labelled with R, we represent it as R; otherwise, let r be the root of t and α the label of r:

- 1. if r has one immediate successor then we represent t as $\langle \alpha, t' \rangle$, where t' is (the representation of) the subtree rooted at the immediate successor of r;
- 2. if r has two immediate successors then we represent t as $\langle t_1, \alpha, t_2 \rangle$, where t_1 and t_2 are (the representations of) the subtrees rooted, respectively, at the left immediate successor and at the right immediate successor of r.

Definition 14 (tree) The tree, $t_{\mathcal{S}}(e)$, and the number of subtrees, $nst_{\mathcal{S}}(e)$, of an \mathcal{S}^c -expression e are defined recursively as follows:

- 1. for all $R \in S$, $t_{\mathcal{S}}(R) = R$ and $nst_{\mathcal{S}}(R) = 1$;
- 2. $t_{\mathcal{S}}(e^{\smile}) = \langle {}^{\smile}, t_{\mathcal{S}}(e) \rangle$ and $nst_{\mathcal{S}}(e^{\smile}) = 1 + nst_{\mathcal{S}}(e);$
- 3. $t_{\mathcal{S}}(e^{\frown}) = \langle \frown, t_{\mathcal{S}}(e) \rangle$ and $nst_{\mathcal{S}}(e^{\frown}) = 1 + nst_{\mathcal{S}}(e);$
 - 24

4. $t_{\mathcal{S}}(e_1 \cap e_2) = \langle t_{\mathcal{S}}(e_1), \cap, t_{\mathcal{S}}(e_2) \rangle$ and $nst_{\mathcal{S}}(e_1 \cap e_2) = 1 + nst_{\mathcal{S}}(e_1) + nst_{\mathcal{S}}(e_2);$ and

5. $t_{\mathcal{S}}(e_1 \circ e_2) = \langle t_{\mathcal{S}}(e_1), \circ, t_{\mathcal{S}}(e_2) \rangle$ and $nst_{\mathcal{S}}(e_1 \circ e_2) = 1 + nst_{\mathcal{S}}(e_1) + nst_{\mathcal{S}}(e_2)$.

Thus the leaves of the tree of an S^c -expression are labelled with elements of S, and the internal nodes with the operators \neg, \cap , \cap and \circ . The number of subtrees, $nst_{\mathcal{S}}(e)$, of an S^c -expression e is the sum of the number, $nl_{\mathcal{S}}(e)$, of leaves of $t_{\mathcal{S}}(e)$ and the number, $no_{\mathcal{S}}(e)$, of operators of e: $nst_{\mathcal{S}}(e) = nl_{\mathcal{S}}(e) + no_{\mathcal{S}}(e)$; $no_{\mathcal{S}}(e)$ indicates the number of internal nodes of $t_{\mathcal{S}}(e)$, i.e., the number of operators to apply in order to get the corresponding element of S^c . The procedure enumerate() in Figure 9 enumerates the closure of a subset S of \mathcal{CYC}_t : the elements of S are supposed ordered. When the procedure completes, variable size indicates the number of relations in the closure of S, the array c contains the elements of the closure of S, and for each $i = 1 \dots size, t[i]$ is the tree of an S^c -expression whose "execution" gives c[i], nst[i] is the number of subtrees of t[i].

Remark 2 In the remainder of the paper, and particularly in the proof of Theorem 9, we refer to the tree t[i] as the tree, $t_{\mathcal{S}}(c[i])$, of the relation c[i] of \mathcal{S}^c , and to nst[i] as the number of subtrees, $nst_{\mathcal{S}}(t_{\mathcal{S}}(c[i]))$, of $t_{\mathcal{S}}(c[i])$: $t[i] = t_{\mathcal{S}}(c[i])$, $nst[i] = nst_{\mathcal{S}}(t_{\mathcal{S}}(c[i]))$.

7 CSPs on cyclic ordering of 2D orientations

We define a \mathcal{CYC}_b -CSP as a CSP of which the constraints are \mathcal{CYC}_b relations on pairs of the variables; a \mathcal{CYC}_t -CSP as a CSP of which the constraints are \mathcal{CYC}_t relations on triples of the variables. For both types of CSPs, the universe is the set 2D \mathcal{O} of 2D orientations. We use the term \mathcal{CYC} -CSP to refer to a CSP which is either a \mathcal{CYC}_b -CSP or a \mathcal{CYC}_t -CSP.

A \mathcal{CYC}_b -matrix (resp. \mathcal{CYC}_t -matrix) of order n is a constraint matrix of order n of which the entries are \mathcal{CYC}_b (resp. \mathcal{CYC}_t) relations. The constraint matrix associated with a \mathcal{CYC}_b -CSP (resp. \mathcal{CYC}_t -CSP) is a \mathcal{CYC}_b -matrix (resp. \mathcal{CYC}_t -matrix).

A scenario of a CYC-CSP is a refinement P' such that all entries of $M^{P'}$ are atoms. A consistent scenario is a scenario which is consistent.

If we make the assumption that a $C\mathcal{YC}$ -CSP does not include the empty constraint, which indicates a trivial inconsistency, then a $C\mathcal{YC}_b$ -CSP is strongly 2-consistent and a $C\mathcal{YC}_t$ -CSP is strongly 3-consistent.

7.1 Achieving path consistency for a CYC_b -CSP

A simple adaptation of Allen's constraint propagation algorithm [1] can be used to achieve path consistency (hence strong 3-consistency) for $C \mathcal{YC}_b$ -CSPs. Applied to a $C \mathcal{YC}_b$ -CSP P, such an adaptation would repeat the following steps

Input: a subset $S = \{R_1, \ldots, R_m\}$ of $C \mathcal{YC}_t$ $(S \subseteq 2^{TER})$. Output: enumeration of the closure \mathcal{S}^c . procedure $enumerate(\mathcal{S}, c, t, nst)$; 1. for $i \leftarrow 1$ to $m\{c[i] \leftarrow R_i; t[i] \leftarrow R_i; nst[i] = 1;\}$ 2. $size \leftarrow m;$ 3. $i \leftarrow 1$; 4. while (i < size)5. $R \leftarrow (c[i])^{\smile};$ 6. if $(R \notin c)$ { $size + +; c[size] \leftarrow R; t[size] \leftarrow (\buildred, t[i]); nst[size] \leftarrow 1 + nst[i];$ } 7. $R \leftarrow (c[i])^{\frown};$ 8. if $(R \notin c)$ { $size + +; c[size] \leftarrow R; t[size] \leftarrow \langle \frown, t[i] \rangle; nst[size] \leftarrow 1 + nst[i];$ } 9. $j \leftarrow 1;$ 10. while $(j \leq i)$ { 11. $R \leftarrow c[i] \cap c[j];$ $\mathrm{if}(R \notin c)\{size + +; c[size] \leftarrow R; t[size] \leftarrow \langle t[i], \cap, t[j] \rangle; nst[size] \leftarrow 1 + c[size] \leftarrow 1 + c[s$ 12.nst[i] + nst[j]; $R \leftarrow c[i] \circ c[j];$ 13. $if(R \notin c)\{size + +; c[size] \leftarrow R; t[size] \leftarrow \langle t[i], \circ, t[j] \rangle; nst[size] \leftarrow 1 +$ 14. nst[i] + nst[j]; $R \leftarrow c[j] \circ c[i];$ 15.16. $if(R \notin c)\{size + +; c[size] \leftarrow R; t[size] \leftarrow \langle t[j], \circ, t[i] \rangle; nst[size] \leftarrow 1 +$ nst[j] + nst[i];17. j + +;} 18. 19. i + +;

Figure 9: Enumeration of the closure of a subset of $C \mathcal{YC}_t$.

20. }



Figure 10: (I) The 'Indian tent'; and (II) its associated $C\mathcal{YC}_b$ -CSP: the CSP is path consistent but not consistent (path consistency does not detect inconsistency even for $C\mathcal{YC}_b$ -CSPs entirely labelled with atoms).

until either stability is reached or the empty relation is detected (indicating inconsistency):

- 1. Consider a triple (X_i, X_j, X_k) of variables verifying $(M^P)_{ij} \not\subseteq ((M^P)_{ik} \circ (M^P)_{kj})$
- 2. $(M^P)_{ij} \leftarrow (M^P)_{ij} \cap (M^P)_{ik} \circ (M^P)_{kj}$
- 3. If $((M^P)_{ij} = \emptyset)$ then exit (the CSP is inconsistent).

Example 5 (the 'Indian tent') The 'Indian tent' consists of a clockwise triangle (ABC), together with a fourth point D which is to the left of each of the directed lines (AB) and (BC) (see Figure 10(I)).

The knowledge about the 'Indian tent' can be represented as a $C\mathcal{YC}_b$ -CSP on four variables, X_1 , X_2 , X_3 and X_4 , representing the orientations of the directed lines (AB), (AC), (BC) and (BD), respectively. From (ABC) being a clockwise triangle, we get a first set of constraints: $\{r(X_2, X_1), r(X_3, X_1), r(X_3, X_2)\}$. From D being to the left of each of the directed lines (AB) and (BC), we get a second set of constraints: $\{l(X_4, X_1), l(X_4, X_3)\}$.

If we add the constraint $r(X_4, X_2)$ to the CSP, which states that the point D should be to the right of the directed line (AC), this leads to an inconsistency. Röhrig [32] has shown that using the CYCORD theory one can detect such an inconsistency, whereas this cannot be detected using classical constraint-based approaches such as those in [8, 9, 18].

The \mathcal{CYC}_b -CSP is represented graphically in Figure 10(II): a \mathcal{CYC}_b constraint R(X,Y) is represented as the directed edge (X,Y) labelled with R. The CSP is path-consistent: $(\forall i, j, k)(P_{ij} \subseteq P_{ik} \circ P_{kj})$.³ However, as mentioned above, the CSP is inconsistent. Therefore:

³This can be easily verified using the \mathcal{CYC}_b composition table.

Theorem 1 Path-consistency does not detect inconsistency even for CYC_b -CSPs entirely labelled with atoms.

7.2 Achieving strong 4-consistency for a CYC_t -CSP

A constraint propagation procedure, s4c(), for \mathcal{CYC}_t -CSPs is given in Figure 11; the procedure is an adaptation of Allen's algorithm [1] to ternary relations. The input is a \mathcal{CYC}_t -CSP P of order n. When the procedure completes, P verifies the following: $(\forall i, j, k, l \leq n)((M^P)_{ijk} \subseteq (M^P)_{ijl} \circ (M^P)_{ilk})$.

The procedure makes use of a queue Queue. Initially, we can assume that all triples (X_i, X_j, X_k) such that $1 \le i \le j \le k \le n$ are entered into Queue. The procedure removes one triple from Queue at a time. When a triple (X_i, X_j, X_k) is removed from Queue, the procedure eventually updates the relations on the neighbouring triples (triples sharing two variables with (X_i, X_j, X_k)). If such a relation is successfully updated, the corresponding triple is sorted, in such a way to have the variable with the smallest index first and the variable with the greatest index last, and the sorted triple is placed in Queue (if it is not already there) since it may in turn constrain the relations on neighbouring triples: this is done by add-to-queue(). The process terminates when Queue becomes empty.

Theorem 2 The constraint propagation procedure s4c() achieves strong 4-consistency for the input CYC_t -CSP, and runs into completion in $O(n^4)$ time, where n is the number of variables of the CSP.

Proof. A \mathcal{CYC}_t -CSP is strongly 3-consistent. Procedure s4c() achieves 4-consistency, therefore it achieves strong 4-consistency. The number of variable triples (X_i, X_j, X_k) is $O(n^3)$. A triple may be placed in Queue at most a constant number of times (24, which is the total number of \mathcal{CYC}_t atoms). Every time a triple is removed from Queue for propagation, the procedure performs O(n) operations.

7.3 A consistent scenario search algorithm for CYC_t -CSPs

We will show that the task of checking consistency for a general $C\mathcal{YC}_t$ -CSP is NP-complete; thus, with the assumption $P \neq NP$, no polynomial algorithm can be found for that task. On the other hand, we will show that the set of $C\mathcal{YC}_t$ atoms is tractable; specifically, we will show that a $C\mathcal{YC}_t$ -CSP for which every three variables X, Y, Z are involved in a constraint of the form t(X, Y, Z), where tis a $C\mathcal{YC}_t$ atom, can be checked for consistency using the s4c() procedure, which performs in polyomial time. We can thus use a backtracking search procedure to check consistency for a general $C\mathcal{YC}_t$ -CSP, which will search for a scenario, if any, of the input $C\mathcal{YC}_t$ -CSP which is strongly 4-consistent, thus consistent. Such a search procedure is provided in Figure 12, which is similar to the one of Ladkin and Reinefeld [21] for temporal interval networks, except that:

Input: a $C \mathcal{Y} C_t$ -CSP P.

Output: the CSP P made strongly 4-consistent.

procedure s4c(P);

- 1. initialise Queue;
- 2. repeat $\{$
- 3. get next triple (X_i, X_j, X_k) from Queue;
- 4. for $m \leftarrow 1$ to $n\{$
- 5. $Temp \leftarrow (M^P)_{ijm} \cap (M^P)_{ijk} \circ (M^P)_{ikm};$
- 6. If $Temp = \emptyset$ then exit (the CSP is inconsistent);
- 7. if $Temp \neq (M^P)_{ijm}$
- 8. {add-to-queue (X_i, X_j, X_m) ; update(P, i, j, m, Temp); }
- 9. $Temp \leftarrow (M^P)_{ikm} \cap (M^P)_{ikj} \circ (M^P)_{ijm};$
- 10. If $Temp = \emptyset$ then exit (the CSP is inconsistent);
- 11. if $Temp \neq (M^P)_{ikm}$
- 12. {add-to-queue (X_i, X_k, X_m) ; update(P, i, k, m, Temp); }
- 13. $Temp \leftarrow (M^P)_{jkm} \cap (M^P)_{jki} \circ (M^P)_{jim};$
- 14. If $Temp = \emptyset$ then exit (the CSP is inconsistent);
- 15. if $Temp \neq (M^P)_{jkm}$
- 16. {add-to-queue (X_j, X_k, X_m) ; update(P, j, k, m, Temp);
- 17. }
- 18. }
- 19. until Queue is empty;

procedure update(P, i, j, k, T);

1.
$$(M^P)_{ijk} \leftarrow T; (M^P)_{ikj} \leftarrow T^{\frown}; (M^P)_{jki} \leftarrow T^{\frown};$$

2. $(M^P)_{jik} \leftarrow ((M^P)_{jki})^{\smile}; (M^P)_{kij} \leftarrow ((M^P)_{jki})^{\frown}; (M^P)_{kji} \leftarrow ((M^P)_{jki})^{\frown}; (M^P)_{kji} \leftarrow$

Figure 11: A constraint propagation procedure for \mathcal{CYC}_t -CSPs.

Input: A CYC_t -CSP P; *Output: true* if and only if *P* is consistent; function consistent(P);1. s4c(P);2. if(P contains the empty relation) return false; 3. else4. if(P contains triples labelled with relations other than atoms)choose such a triple, say (X_i, X_j, X_k) ; 5. $T \leftarrow (M^P)_{ijk};$ 6. for each atom t in $T\{$ 7. refine $(M^P)_{ijk}$ to t (i.e., $(M^P)_{ijk} \leftarrow t)$; 8. 9. if(consistent(P))return true; 10. } $(M^P)_{ijk} \leftarrow T;$ 11. 12.return false; 13.} 14.else return true;

Figure 12: A consistent scenario search algorithm for \mathcal{CYC}_t -CSPs.

- 1. it refines the relation on a triple of variables at each node of the search tree, instead of the relation on a pair of variables; and
- 2. it makes use of the procedure s4c(), which achieves strong 4-consistency, in the preprocessing step and as the filtering method during the search, instead of a path consistency procedure.

The other details are similar to those of Ladkin and Reinefeld's algorithm.

Definition 15 Let P denote a CYC_t -CSP of order n:

- 1. P is projectable if for all $i, j, k, (M^P)_{ijk}$ is a projectable CYC_t relation.
- 2. The projection of P is the \mathcal{CYC}_b - $CSP \bigtriangledown (P)$ with the same set of variables, and such that: $(\forall i, j \leq n)((M^{\bigtriangledown (P)})_{ji} = \bigcap_{k \leq n} [\bigtriangledown^1((M^P)_{ijk}) \cap \bigtriangledown^2((M^P)_{kij}) \cap$
 - $\bigtriangledown^3((M^P)_{ikj})]).$

The next two theorems will be needed in the next section, for the proof of Theorem 6.

Theorem 3 A projectable CYC_t -CSP is equivalent to its projection.

Proof. Let *P* be a projectable $C \mathcal{YC}_t$ -CSP; thus:

$$(\forall i, j, k)[(M^P)_{ijk} \equiv \Pi(\bigtriangledown^1((M^P)_{ijk}), \bigtriangledown^2((M^P)_{ijk}), \bigtriangledown^3((M^P)_{ijk}))]$$

In other words, the constraint $(M^P)_{ijk}(X_i, X_j, X_k)$ can be equivalently written as the following conjunction of binary constraints: $\nabla^1((M^P)_{ijk})(X_j, X_i) \wedge \nabla^2((M^P)_{ijk})(X_k, X_j) \wedge \nabla^3((M^P)_{ijk})(X_k, X_i)$. *P* can be written as the conjunction $\bigwedge_{i,j,k \leq n} (M^P)_{ijk}(X_i, X_j, X_k)$; replacing the constraint $(M^P)_{ijk}(X_i, X_j, X_k)$

by the equivalent conjunction of binary constraints, we get:

$$P \equiv \bigwedge_{i,j,k \leq n} [\bigtriangledown^1((M^P)_{ijk})(X_j,X_i) \land \bigtriangledown^2((M^P)_{ijk})(X_k,X_j) \land \bigtriangledown^3((M^P)_{ijk})(X_k,X_i)]$$

Because the conjunction considers all possible triples (i, j, k), with $i, j, k \leq n$, it can be split into:

$$P \equiv \bigwedge_{i,j,k \leq n} \bigtriangledown^1 ((M^P)_{ijk})(X_j, X_i) \land \bigwedge_{i,j,k \leq n} \bigtriangledown^2 ((M^P)_{ijk})(X_k, X_j) \land \bigwedge_{i,j,k \leq n} \bigtriangledown^3 ((M^P)_{ijk})(X_k, X_i) \land \bigwedge_{i,j,k \leq n} \bigtriangledown ((M^P)_{ijk})(X_k, X_i) \land \bigwedge_{i,j,k \leq n} ((M^P)_{ijk})(X_k, X_i) \land \bigwedge_{i,j,k \in n} ((M^P)_{ijk})(X_k, X_i) \land \bigwedge_{i,j,k \in$$

We consider now the main three subconjunctions and rename i, j, k as k, i, j, respectively, in the second subconjunction, and as i, k, j, respectively, in the third subconjunction; we get:

$$P \equiv \bigwedge_{i,j,k \leq n} \bigtriangledown^1 ((M^P)_{ijk})(X_j, X_i) \land \bigwedge_{i,j,k \leq n} \bigtriangledown^2 ((M^P)_{kij})(X_j, X_i) \land \bigwedge_{i,j,k \leq n} \bigtriangledown^3 ((M^P)_{ikj})(X_j, X_i) \land \bigwedge_{i,j,k \leq n} ((M^P)_{ikj})(X_j, X_i) \land \bigwedge_{i,j,k \in n} ((M^P)_{ikj})(X_j) \land \bigwedge_{i,j,k \in n} ((M^P)_{ikj})(X_j) \land \bigwedge_{i,j,k \in n} ((M^P)_{ikj})(X_j) \land \bigwedge_{i,j,k \in n} ((M^P$$

Putting back the three subconjunctions into one main conjunction, we get:

$$P \equiv \bigwedge_{i,j,k \leq n} [\bigtriangledown^1((M^P)_{ijk})(X_j, X_i) \land \bigtriangledown^2((M^P)_{kij})(X_j, X_i) \land \bigtriangledown^3((M^P)_{ikj})(X_j, X_i)]$$

which is equivalent to:

$$P\equiv igwedge _{i,j\leq n} [igwedge _{k\leq n} \bigtriangledown^1 ((M^P)_{ijk})(X_j,X_i) \wedge \bigtriangledown^2 ((M^P)_{kij})(X_j,X_i) \wedge \bigtriangledown^3 ((M^P)_{ikj})(X_j,X_i)]$$

which in turn is equivalent to:

$$P \equiv \bigwedge_{i,j \leq n} \left[\bigcap_{k \leq n} [\nabla^1((M^P)_{ijk}) \wedge \nabla^2((M^P)_{kij}) \wedge \nabla^3((M^P)_{ikj})](X_j, X_i) \right]$$

This corresponds exactly to the constraint matrix of the projection $\nabla(P)$ of P.

Theorem 4 Let P denote a projectable \mathcal{CYC}_t -CSP of order n. If P is strongly 4-consistent then its projection $\nabla(P)$ verifies the following: $(\forall i, j, k_1, k_2 \leq n)[(M^{\nabla(P)})_{ji} = \nabla^1((M^P)_{ijk_1}) = \nabla^2((M^P)_{k_1ij}) = \nabla^3((M^P)_{ik_1j}) = \nabla^1((M^P)_{ijk_2})].$

Proof (sketch). Strong 4-consistency of P implies its closure under the operations of converse and rotation, as well as under what we will refer to as the operation of strong 4-consistency, or s4c-operation for short:

$$(M^P)_{ijk} \leftarrow (M^P)_{ijk} \cap (M^P)_{ijl} \circ (M^P)_{ilk}$$

From the closure under the operations of converse and rotation, we get:

$$(\forall b_1 b_2 b_3 \in TER)$$

 $(b_1b_2b_3 \in (M^P)_{ijk_1} \Leftrightarrow b_3(b_2)^{\smile}b_1 \in (M^P)_{ik_1j} \Leftrightarrow (b_3)^{\smile}b_1(b_2)^{\smile} \in (M^P)_{k_1ij})$

which implies the following:

$$(\forall b_1 \in BIN)[b_1 \in \bigtriangledown^1((M^P)_{ijk_1}) \Leftrightarrow b_1 \in \bigtriangledown^3((M^P)_{ik_1j}) \Leftrightarrow b_1 \in \bigtriangledown^2((M^P)_{k_1ij})]$$

Thus $(\forall i, j, k_1) [\bigtriangledown^{(MP)}_{ijk_1}) = \bigtriangledown^2 ((M^P)_{k_1 ij}) = \bigtriangledown^3 ((M^P)_{ik_1 j})].$

Let $b_1 \in \nabla^1((M^P)_{ijk_1})$, and suppose that for some $k_2 \neq k_1, b_1 \notin \nabla^1((M^P)_{ijk_2})$. We use the fact that given two \mathcal{CVC}_t atoms t_1 and t_2 and a \mathcal{CVC}_b atom b, if $b \notin \nabla^1(t_1)$ then $b \notin \nabla^1(t_1 \circ t_2)$. We get that $b_1 \notin \nabla^1((M^P)_{ijk_2} \circ (M^P)_{ik_2k_1})$. Now closure under the *s*4*c*-operation implies that $(M^P)_{ijk_1} \subseteq (M^P)_{ijk_2} \circ (M^P)_{ik_2k_1}$. From $b_1 \notin \nabla^1((M^P)_{ijk_2} \circ (M^P)_{ik_2k_1})$, we derive that $b_1 \notin \nabla^1((M^P)_{ijk_1})$, which contradicts our supposition.

A $C \mathcal{YC}_b$ -CSP P can be transformed into an equivalent $C \mathcal{YC}_t$ -CSP, say P', as follows:

- 1. P' has the same set of variables as P; and
- 2. $(\forall i, j, k)((M^{P'})_{ijk} = \Pi((M^P)_{ji}, (M^P)_{kj}, (M^P)_{ki}).$



Figure 13: Illustration of the proof of Theorem 6.

8 A tractability result

The aim of this section is to show that the closure under strong 4-consistency, $(\mathcal{AT}_t)^{s4c}$, of the set $\mathcal{AT}_t = \{\{t\} : t \in TER\}$ of all \mathcal{CVC}_t atoms is tractable; more specifically, using the terminology in [2], we show that the \mathcal{CVC}_t composition tables are complete for $(\mathcal{AT}_t)^{s4c}$. We first prove that if a \mathcal{CVC}_t -CSP expressed in TCH is strongly 4-consistent then it is globally consistent, from which the result will follow. The proof will need Helly's convexity theorem:

Theorem 5 (Helly's Theorem [4]) Let S be a set of convex regions of the n-dimensional space \mathbb{R}^n . If every n + 1 elements in S have a non empty intersection then the intersection of all elements of S is non empty.

For n = 2, the theorem states that if a set of convex planar regions is such that every three regions in the set have a non empty intersection then the intersection of all regions in the set is non empty.

Van Beek [36] has used the specialisation to n = 1 of Helly's theorem to prove a tractability result for path consistent CSPs of Allen's convex relations. We will need the specialisation to n = 2.

Theorem 6 Let P be a CYC_t -CSP expressed in TCH: $(\forall i, j, k)((M^P)_{ijk} \in TCH)$. If P is strongly 4-consistent then it is globally consistent.

Proof. Since P is expressed in TCH and is strongly 4-consistent, we have the following:

- 1. *P* is equivalent to its projection $\nabla(P)$, which is a \mathcal{CYC}_b -CSP expressed in BCH: $(\forall i, j)((M^{\nabla(P)})_{ij} \in BCH)$.
- 2. The projection $\bigtriangledown(P)$ is strongly 4-consistent.

So the problem becomes that of showing that $\bigtriangledown(P)$ is globally consistent. For this purpose, we suppose that the instantiation $(X_{i_1}, X_{i_2}, \ldots, X_{i_k}) = (z_1, z_2, \ldots, z_k), k \geq 4$, is a solution to the k-variable sub-CSP $(\bigtriangledown(P))_{|\{X_{i_1}, \ldots, X_{i_k}\}}$

of $\nabla(P)$. We need to prove that the partial solution can be extended to any (k+1)st variable, say $X_{i_{k+1}}$, of $\nabla(P)$.⁴ This is equivalent to showing that the following sectors have a non empty intersection (see Figure 13 for illustration): $sect(z_1, (M^{\bigtriangledown(P)})_{i_{k+1}i_1}), sect(z_2, (M^{\bigtriangledown(P)})_{i_{k+1}i_2}), \dots, sect(z_k, (M^{\bigtriangledown(P)})_{i_{k+1}i_k}).$ Since the $(M^{\bigtriangledown(P)})_{i_{k+1}i_j}, j = 1 \dots k$, belong to *BCH*, each of these sectors

is:

- 1. a convex subset of the plane; or
- 2. almost equal to the surface of circle $\mathcal{C}_{O,1}$ (its topological closure is equal to that surface).

We split these sectors into those verifying condition (1) and those verifying condition (2). We assume, without loss of generality, that the first m verify condition (1), and the last k - m verify condition (2). We write the intersection of the sectors as $I = I_1 \cap I_2$, with $I_1 = \bigcap_{j=1}^m sect(z_j, (M^{\bigtriangledown (P)})_{i_{k+1}i_j}), I_2 =$ $\bigcap_{j=m+1}^k sect(z_j, (M^{\bigtriangledown (P)})_{i_{k+1}i_j}).$

Due to strong 4-consistency, every three of these sectors have a non empty intersection. If any of the sectors is a radius (the corresponding relation is either e or o) then the entire intersection must be equal to that radius since the sector intersects with every other two.

We now need to show that when no sector reduces to a radius, the intersection is still non empty:

Case 1: m=k

This means that all sectors are convex. Since every three of them have a non empty intersection, Helly's theorem immediately implies that the intersection of all sectors is non empty.

Case 2: m=0

This means that no sector is convex; which in turn implies that each sector is such that its topological closure covers the entire surface of $\mathcal{C}_{O,1}$. Hence, for all $j = 1 \dots k$:

- 1. the sector $sect(z_j, (M^{\nabla(P)})_{i_{k+1}i_j})$ is equal to the entire surface of $\mathcal{C}_{O,1}$ minus the centre (the relation $(M^P)_{i_{k+1}i_j}$ is equal to BIN); or
- 2. the sector $sect(z_j, (M^{\nabla(P)})_{i_{k+1}i_j})$ is equal to the entire surface of $\mathcal{C}_{O,1}$ minus the centre and one or two radii (the relation $(M^P)_{i_{k+1}i_i}$ is equal to $\{e, l, r\}, \{l, o, r\} \text{ or } \{l, r\}$.

⁴Since the $C \mathcal{YC}_t$ -CSP P is projectable, any solution to any sub-CSP of the projection $\nabla(P)$ is solution to the corresponding sub-CSP of P. This would not be necessarily the case if Pwere not projectable.

So the intersection of all sectors is equal to the entire surface of $C_{O,1}$ minus the centre and a finite number (at most 2k) of radii. Since the surface is of dimension 2, a radius of dimension 1, and the centre of dimension 0, the intersection must be non empty (of dimension 2).

Case 3: 0 < m < k

This means that some sectors (at least one) are convex, the others (at least one) are such that their topological closures cover the entire surface of $C_{O,1}$. The intersection I_1 is non empty due to Helly's theorem, since every three sectors appearing in it have a non empty intersection. We need to consider two subcases.

Subcase 3.1: I_1 is a single radius, say t

Since no sector reduces to a radius, and the sectors appearing in I_1 are less than π , there must exist two sectors, say s_1 and s_2 , appearing in I_1 such that their intersection is t. Since, due to strong 4-consistency, s_1 and s_2 together with any sector appearing in I_2 form a non empty intersection, the whole intersection, i.e., I, must be equal to t.

Subcase 3.2: I_1 is a 2-dimensional (convex) sector

The intersection I_2 is the entire surface of $C_{O,1}$ minus the centre and a finite number (at most 2(k-m)) of radii. Since the centre is of dimension 0, a finite union of radii is of dimension 0 or 1, and the intersection I_1 is of dimension 2, the whole intersection I must be non empty (of dimension 2).

The intersection of all sectors is non empty in all cases. The partial solution can therefore be extended to variable $X_{i_{k+1}}$ (which can be instantiated with any orientation in the intersection of the k sectors).

It follows from Theorems 2 and 6 that if the TCH subclass is closed under strong 4-consistency, it must be tractable. Unfortunately, as illustrated by the following example, TCH is not so closed.

Example 6 (non closure of *TCH* under strong 4-consistency) The \mathcal{CYC}_b - *CSP* depicted in Figure 14 can be represented as the projectable \mathcal{CYC}_t -*CSP* Pverifying the following: $(M^P)_{123} = lll, (M^P)_{124} = \Pi(l, \{l, r\}, \{l, r\}), (M^P)_{134} =$ $P_{234} = \Pi(l, l, \{l, r\})$. Applying the propagation procedure s4c() to P leaves unchanged $(M^P)_{123}, (M^P)_{134}, (M^P)_{234}$, but transforms $(M^P)_{124}$ into the relation $\{lll, llr, lrr\}$, which is not projectable: this is done by the operation $(M^P)_{124} \leftarrow$ $(M^P)_{124} \cap (M^P)_{123} \circ (M^P)_{134}$.

Indeed, as we will show, the subset TCH is not tractable. Even worse, we will prove that the strict subset $\mathcal{AT}_t \cup \{TER\}$ is already NP-complete (Corollary 4).



Figure 14: Illustration of non closure of TCH under strong 4-consistency.

Ø		Inc	1	nlm	1	$\Pi(f \mid o \mid n \mid l \mid n)$	
V		110		111		$\Pi(\{i, 0, i\}, i, i)$	$\prod (r \{l \ r\} \ r)$
eee		lrl		rol		$\Pi(\{e, l, r\}, l, l)$	
11	-	,		1		$\Pi(I)$	$11(l, l, \{l, r\})$
eu		irr		rri		$\Pi(\{l,r\},l,r)$	$\Pi(l \in \{a, l, m\})$
e00		0e 0	1	rro		$\Pi(\{l \ r\} \ r \ r)$	$\Pi(i, i, \chi e, i, i f)$
000	_			11.0		$\Pi([i,i],i,i)$	$\Pi(r, \{l, r\}, l)$
err		olr		rrr		$\Pi(\{l, o, r\}, r, l)$	
lal		000		$\Pi(l (a l m) l)$		$\Pi([l_n]_n, l)$	$\Pi(l, \{l, r\}, l)$
iei		000		$\Pi(i, \{e, i, r\}, i)$		$\Pi(\{i,r\},r,i)$	$\Pi(r \ r \ \{l \ r\})$
111		orl		$\Pi(l, \{l, o, r\}, r)$		$\Pi(\{l, r\}, l, l)$	$\mathbf{H}(r, r, [r, r])$
	-	0.0		==((, ((, (, (, (, (), ())))))))		$\Xi((i,i,j,i,i))$	$\Pi(l, \{l, r\}, r)$
llo		rer		$\Pi(r, \{e, l, r\}, r)$		$\Pi(r, l, \{e, l, r\})$	$\pi(n, l, (l, n))$
11	-	ml a	ł	$\Pi(n(l, n))$		$\Pi(m, m, (l, a, m))$	$\Pi(r, l, \{l, r\})$
117		rie		$\Pi(r, \{i, 0, r\}, i)$		$\Pi(r, r, \{l, 0, r\})$	$\Pi(l r \{l r\})$
lor		rll		$\Pi(\{e, l, r\}, r, r)$		$\Pi(l, l, \{l, o, r\})$	11(i, i, [i, i])
	1		1	**((°,°,′))/////		(•,•, (•, •, •, · J)	

Figure 15: Enumeration of $(\mathcal{AT}_t)^{s4c}$.

The set $(\mathcal{AT}_t)^{s4c}$ includes all 28 entries of the \mathcal{CYC}_t composition tables: the 24 atoms together with the relations $\Pi(l, \{e, l, r\}, l), \Pi(l, \{l, o, r\}, r), \Pi(r, \{e, l, r\}, r), \Pi(r, \{l, o, r\}, l)$. Furthermore, enumerating $(\mathcal{AT}_t)^{s4c}$ leads to 49 relations (including the empty relation), all of which are {convex,holed} relations (belong to TCH). This immediately gives the following corollary, stating tractability of $(\mathcal{AT}_t)^{s4c}$.

Corollary 1 (tractability of $(\mathcal{AT}_t)^{s4c}$) Let P be a \mathcal{CYC}_t -CSP expressed in $(\mathcal{AT}_t)^{s4c}$: $(\forall i, j, k)((M^P)_{ijk} \in (\mathcal{AT}_t)^{s4c})$. Deciding consistency for P is tractable.

Proof. Immediate from Theorems 2 and 6. The enumeration of $(\mathcal{AT}_t)^{s4c}$ is given in Figure 15.

Example 7 Transforming the \mathcal{CYC}_b -CSP of the 'Indian tent' into a \mathcal{CYC}_t -CSP, say P', leads to $(M^{P'})_{123} = rrr, (M^{P'})_{124} = rrl, (M^{P'})_{134} = rll, (M^{P'})_{234} = rlr. P'$ lies in $(\mathcal{AT}_t)^{s4c}$, hence the propagation procedure s4c() must detect

its inconsistency. Indeed, the operation $(M^{P'})_{124} \leftarrow (M^{P'})_{124} \cap (M^{P'})_{123} \circ (M^{P'})_{134}$ leads to the empty relation, since $rrr \circ rll = rll$.

Corollary 2 (tractability of \mathcal{AT}_b) Let P be a \mathcal{CYC}_b -CSP expressed in \mathcal{AT}_b : $(\forall i, j)((M^P)_{ij} \in \mathcal{AT}_b)$. Deciding consistency for P is tractable.

Proof (sketch). Let P be a \mathcal{CYC}_b -CSP as stated in the corollary. Construct from P the \mathcal{CYC}_t -CSP P' of which P is the projection: $(\forall i, j, k)((M^{P'})_{ijk} = \Pi((M^P)_{ji}, (M^P)_{kj}, (M^P)_{ki}))$. All entries of M^P belong to \mathcal{AT}_t . From Corollary 1, deciding consistency for P', thus for P, is tractable.

9 Intractability results

This section presents some intractability results:

- 1. We first show that the RA $C \mathcal{YC}_t$ is NP-complete; this directly follows from the NP-completeness of the CYCORD theory [15].
- 2. We show that the weak closure $(PAR)^{wc}$ of the subset $PAR = \{\{oeo, ooe\}, \{eee, eoo, ooe\}, \{eee, eoo, oeo, ooe\}\}$ of \mathcal{CYC}_t , which expresses only information on parallel orientations, is NP-complete. This gives an idea of how hard reasoning within \mathcal{CYC}_t is: even if we restrict ourselves to a world of parallel orientations, reasoning within that world is already NP-complete!
- 3. We show that provided that a subset S of $C\mathcal{YC}_t$ includes the relations *eee* and *TER*, deciding consistency for a CSP expressed in S^c can be polynomially reduced to deciding consistency for a CSP expressed in S.
- 4. We use the previous result to prove that the set $\mathcal{AT}_t^+ = \mathcal{AT}_t \cup \{TER\}$, i.e., the set obtained by adding the universal relation to the set of all \mathcal{CYC}_t atoms, is NP-complete.
- 5. From NP-completeness of \mathcal{AT}_{t}^{+} , we derive NP-completeness of $\mathcal{AT}_{b}^{+} = \mathcal{AT}_{b} \cup \{BIN\}$, thus of the RA \mathcal{CYC}_{b} .

Theorem 7 Deciding consistency of a CYC_t -CSP is NP-complete.

Proof: The set \mathcal{AT}_t of all \mathcal{CYC}_t atoms is tractable (Corollary 1); thus, if a \mathcal{CYC}_t -CSP is such that on every triple (X, Y, Z) there is a constraint of the form t(X, Y, Z), where t is an atom, deciding its consistency is polynomial, and can be achieved using the s4c() procedure. Therefore, all we need to show is that there exists a deterministic polynomial transformation of an instance of an NP-complete problem to a \mathcal{CYC}_t -CSP [16].

The CYCORD theory is NP-complete [15]. The transformation of a problem expressed in the CYCORD theory (a conjunction of CYCORD relations) into a

problem expressed in $C\mathcal{YC}_t$ (i.e., into a $C\mathcal{YC}_t$ -CSP) is immediate from the rule illustrated in Figure 8(top) (see Example 2(3)) transforming a CYCORD relation into a $C\mathcal{YC}_t$ relation. Specifically, such a problem, say P, can be transformed into a $C\mathcal{YC}_t$ -CSP, say P', in the following way:

- 1. Initialise all entries of $M^{P'}$ to the universal \mathcal{CYC}_t relation TER: $(\forall i, j, k)((M^{P'})_{ijk} \leftarrow TER)$
- 2. Initialise the diagonal elements to eee: $(\forall i)((M^{P'})_{iii} \leftarrow eee);$
- For all CYCORD relation X_i-X_j-X_k of P, stating that orientations X_i, X_j, X_k are distinct from each other and encountered in that order when we turn in a clockwise direction starting from X_i, perform the following: T ← (M^{P'})_{ijk} ∩ cyc; update(P', i, j, k, T);

The procedure update() is defined in Figure 11, just after the procedure s4c(). By construction, $M^{P'}$ is a constraint matrix over \mathcal{CYC}_t . The transformation is deterministic and polynomial, and P is satisfiable if and only if P' is consistent.

Corollary 3 Let P be a CYC_t -CSP expressed in $CYCORD^+ = \{cyc, cyc^{\checkmark}, eee, TER\}$: $(\forall i, j, k)((M^P)_{ijk} \in CYCORD^+)$. Deciding consistency for P is NP-complete.

Proof (sketch). In the proof of Theorem 7, the $C\mathcal{YC}_t$ -CSP P' associated with a problem P expressed in the CYCORD theory is such that its constraint matrix $M^{P'}$ is entirely expressed in $\{cyc, cyc^{\frown}, eee, TER\}$ (we notice that the set $\{cyc, cyc^{\frown}, eee, TER, \emptyset\}$ is closed under intersection, rotation, and converse).

The weak closure of the set $PAR = \{\{oeo, ooe\}, \{eee, eoo, ooe\}, \{eee, eoo, oeo, ooe\}\}$ contains 15 of the 16 elements of $2^{\{eee, eoo, oeo, ooe\}}$; it can be easily enumerated:

Theorem 8 (NP-completeness of $(PAR)^{wc}$) Let P be a CYC_t -CSP expressed in $(PAR)^{wc}$: $(\forall i, j, k)((M^P)_{ijk} \in (PAR)^{wc})$. Deciding consistency for P is NP-complete.

Proof. The subset $(PAR)^{wc}$ belongs to NP, since solving a \mathcal{CYC}_t -CSP of atoms is polynomial (Corollary 1). We need to prove that there exists a (deterministic) polynomial transformation of an instance of an NP-complete problem (we consider an instance of 3-SAT: a instance of SAT of which every clause contains exactly three literals) into a \mathcal{CYC}_t -CSP expressed in $(PAR)^{wc}$ in such a way that the former is satisfiable (has a model) if and only if the latter is consistent.

Suppose that S is an instance of 3-SAT, and denote by:

- 1. $Lit(S) = \{\ell_1, \ldots, \ell_n\}$ the set of literals appearing in S;
- 2. Cl(S) the set of clauses of S; and
- 3. BinCl(S) the set of binary clauses which are subclauses of clauses in Cl(S).

The \mathcal{CYC}_t -CSP, P_S , we associate with S is as follows. Its set of variables is $V = \{X_c | c \in Lit(S) \cup BinCl(S)\} \cup \{X_0\}$. X_0 is a truth determining variable: all orientations which are equal to X_0 correspond to elements of $Lit(S) \cup BinCl(S)$ which are true, the others (those which are opposite to X_0) to elements of $Lit(S) \cup BinCl(S)$ which are false. The constraint matrix of P_S , M^{P_S} , is regarded as being indexed with elements from $\{0\} \cup Lit(S) \cup BinCl(S)$, and the entry $(M^{P_S})_{abc}$ stands for the relation on triple (X_a, X_b, X_c) :

- 1. Initialise all entries of M^{P_s} to {eee, eoo, oeo, ooe}: $(\forall a, b, c)((M^{P_s})_{abc} \leftarrow \{eee, eoo, oeo, ooe\})$
- 2. Initialise the diagonal elements to eee: $(\forall a)((M^{P_S})_{aaa} \leftarrow eee);$
- 3. for all pairs $(X_p, X_{\overline{p}})$ of variables such that $\{p, \overline{p}\} \subseteq Lit(S)$, p and \overline{p} should have complementary truth values; hence X_p and $X_{\overline{p}}$ should be opposite to each other in P_S : $T \leftarrow (M^{P_S})_{0\overline{p}p} \cap \{eoo, ooe\}; update(P_S, 0, \overline{p}, p, T);$
- 4. for all variables X_{c_1}, X_{c_2} such that $(c_1 \vee c_2)$ is a clause of S, c_1 and c_2 cannot be simultaneously false; translated into P_S, X_{c_1} and X_{c_2} should not be both opposite to $X_0: T \leftarrow (M^{P_S})_{0c_1c_2} \cap \{eee, eoo, ooe\}; update(P_S, 0, c_1, c_2, T);$
- 5. for all variables $X_{(\ell_1 \vee \ell_2)}, X_{\ell_1}$, if ℓ_1 is true then so is $(\ell_1 \vee \ell_2)$; translated into P_S, X_0 and X_{ℓ_1} should not be both opposite to $X_{(\ell_1 \vee \ell_2)}$: $T \leftarrow (M^{P_S})_{(\ell_1 \vee \ell_2)\ell_{10}} \cap \{eee, eoo, ooe\}; update(P_S, \ell_1 \vee \ell_2, \ell_1, 0, T);$

Again, the procedure update() is defined in Figure 11, just after the procedure s4c(). The transformation is deterministic and polynomial. Moreover, since $(PAR)^{wc}$ is closed under intersection, converse and rotation, the final matrix M^{P_S} is a constraint matrix over $(PAR)^{wc}$. If M is a model of S, it is mapped to a solution of P_S as follows. X_0 is assigned any value of $[0, 2\pi)$. For all $\ell \in Lit(S)$, X_ℓ is assigned the same value as X_0 if M assigns the value true to literal ℓ , the value opposite to that of X_0 otherwise. For all $(\ell_1 \vee \ell_2) \in BinCl(S), X_{(\ell_1 \vee \ell_2)}$ is assigned the same value as X_0 if either X_{ℓ_1} or X_{ℓ_2} is assigned the same value as X_0 if either X_{ℓ_1} or X_{ℓ_2} is assigned the same value otherwise. On the other hand, any solution to P_S can be mapped to a model of S by assigning to every literal ℓ the value true if and only if the variable X_ℓ is assigned the same value as X_0 .

Before going further in the presentation of our intractability results, we want to be clear with respect to the issue of representing a $C \mathcal{YC}_t$ -CSP. The most convenient way for representing such a CSP is certainly the use of an $n \times n \times n$ matrix, where n is the order of the CSP; one reason for this is that the standard

way for constraint propagation algorithms and for solution search algorithms, which constitute the main reasoning tools for constraint-based frameworks, to deal with a CSP is to have it represented as a matrix. We have assumed so far that the matrix associated with a $C\mathcal{YC}_t$ -CSP was a constraint matrix; i.e., it verifies the diagonal property, the converse property, and the rotation property. However, in terms of solutions, if we associate with a $C\mathcal{YC}_t$ -CSP P the $n \times n \times n$ -matrix $M^{P,2}$ defined as follows:

- 1. Initialise all entries to the universal relation TER: $(\forall i, j, k)((M^{P,2})_{ijk} \leftarrow TER)$
- 2. Initialise all diagonal elements to eee: $(\forall i)((M^{P,2})_{iii} \leftarrow eee)$
- 3. For all triples (X_i, X_j, X_k) of variables such that a constraint $R(X_i, X_j, X_k)$ is specified: $(M^{P,2})_{ijk} \leftarrow (M^{P,2})_{ijk} \cap R$

then the matrices M^P and $M^{P,2}$ are equivalent, i.e., they have the same set of solutions.

Binary CSPs of Allen's relations on pairs of interval variables in which every two variables are involved in exactly one constraint are called **normalised sets** of interval formulas in [30].

Definition 16 An orientation formula is a CYC_t relation on a triple of variables, i.e., a constraint of the form R(X, Y, Z), where R is a CYC_t relation. A normalised set of orientation formulas is a CYC_t -CSP given as a set of constraints in which every three variables are involved in exactly one constraint.

Given a \mathcal{CYC}_t -CSP P, the matrix M^P is closed under the operations of converse and rotation; this is not necessarily the case for the matrix $M^{P,2}$: in particular, if P is a normalised set of orientation formulas then for any three variables X_i , X_j and X_k , at most one element in the set $\{(M^{P,2})_{lmn} : \{l, m, n\} = \{i, j, k\}\}$ is not the universal relation.

Remark 3 If a subset S of CYC_t includes the relations even and TER then a normalised set of orientation formulas, say P, which is entirely expressed in S is such that its associated matrix $M^{P,2}$ is also entirely expressed in S.

Theorem 9 Let S be a subset of CYC_t such that $ee \in S$ and $TER \in S$. Deciding consistency for a normalised set of orientation formulas expressed in S^c can be polynomially reduced to deciding consistency for a normalised set of orientation formulas expressed in S.

Proof. We have seen how, given a subset S of \mathcal{CYC}_t , to associate with each relation R in the closure S^c a tree $t_S(R)$ in such a way that the "execution" of $t_S(R)$ gives R itself (see Definition ?? and Remark 2). We use the tree of a relation in S^c to transform a normalised set of orientation formulas expressed in S^c into an equivalent set of orientation formulas, g(P), expressed in S and in which every three variables are involved in at most one constraint:

- 1. $g({R(X, Y, Z)}) = h(t_{\mathcal{S}}(R)(X, Y, Z))$, for all $R \in \mathcal{S}^{c}$
- 2. $g(\{R(X,Y,Z)\} \cup P') = g(\{R(X,Y,Z)\}) \cup g(P')$, where $R \in S^c$ and P' is a non empty set of orientation formulas expressed in S^c .

The mapping h is defined as follows:

- 1. $h(R(X, Y, Z)) = \{R(X, Y, Z)\}, \text{ for all } R \in \mathcal{S}$
- 2. $h(\langle , t \rangle(X, Y, Z)) = h(t(X, Z, Y))$
- 3. $h(\langle \frown, t \rangle(X, Y, Z)) = h(t(Z, X, Y))$
- 4. $h(\langle t_1, \cap, t_2 \rangle (X, Y, Z)) = h(t_1(X, Y, Z)) \cup h(t_2(X, Y, Z')) \cup \{eee(Z, Z', Z)\},$ where Z' is a fresh variable
- 5. $h(\langle t_1, \circ, t_2 \rangle(X, Y, Z)) = h(t_1(X, Y, W)) \cup h(t_2(X, W, Z))$, where W is a fresh variable.

By construction, g(P) is a set of orientation formulas expressed in S with the property that every three variables are involved in at most one formula. g(P) is transformed into an equivalent normalised set of orientation formulas by creating for every three variables X, Y, Z not already involved in any formula the formula TER(X, Y, Z).

Let *m* denote the number of orientation formulas in *P*, and refer to the orientation formulas as $f_1(X_{1_1}, X_{1_2}, X_{1_3}), \ldots, f_i(X_{i_1}, X_{i_2}, X_{i_3}), \ldots, f_m(X_{m_1}, X_{m_2}, X_{m_3})$. For each $i = 1 \ldots m$, let s_i denote the size of f_i , i.e., the number, $nst_{\mathcal{S}}(f_i)$, of subtrees of f_i (see Remark 2): $s_i = nst_{\mathcal{S}}(f_i)$. If *s* is the greatest of the s_i 's then the construction takes O(ms) time; the transformation is thus polynomial.

We are now in a position to derive that for both of the presented RAs, CYC_b and CYC_t , we "jump" from tractability to intractability if add the universal relation to the set of all atoms.

Corollary 4 The subset $\mathcal{AT}_t^+ = \mathcal{AT}_t \cup \{TER\}$ of \mathcal{CYC}_t is NP-complete.

Proof. Any set of orientation formulas over \mathcal{AT}_t^+ can be converted into an equivalent normalised set of orientation formulas over \mathcal{AT}_t^+ . The subset \mathcal{AT}_t^+ includes the relations *eee* and *TER*. From Corollary 3 and Theorem 9, and the closure of $(\mathcal{AT}_t^+)^c$ under converse, it is sufficient to show that the relation *cyc* belongs to $(\mathcal{AT}_t^+)^c$. The following sequence shows that this is indeed the case:

- 1. $R_1 = lll, R_2 = llr, R_3 = rll$
- 2. $R_4 = R_1 \circ R_2 = \{llr, lor, lrr\}$

3.
$$R_5 = (R_4)^{\frown} = \{llr, olr, rlr\}$$

- 4. $R_6 = R_5 \circ R_3 = \{lrl, orl, rll, rol, rrl\}$
 - 41

- 5. $R_7 = (R_6)^{\frown} = \{lrl, orl, rrl, rro, rrr\}$
- 6. $R_8 = R_7 \circ R_7 = \{lrl, orl, rll, rol, rrl, rro, rrr\} = cyc. \blacksquare$

Corollary 5 The subset $\mathcal{AT}_b^+ = \mathcal{AT}_b \cup \{BIN\}$ of \mathcal{CYC}_b is NP-complete.

Proof. A \mathcal{CYC}_b -CSP of atoms can be solved in polynomial time (Corollary 2). Thus we need to show that there is a polynomial deterministic transformation of an instance of an NP-complete problem into a problem expressed in \mathcal{AT}_b^+ . We consider a normalised set, P, of orientation formulas expressed in \mathcal{AT}_t^+ . According to Corollary 4, deciding consistency for P is NP-complete. The set $\mathcal{AT}_t^+ \cup \{\emptyset\}$ is closed under converse, rotation and intersection (in other words, $(\mathcal{AT}_t^+ \cup \{\emptyset\})^{wc} = \mathcal{AT}_t^+ \cup \{\emptyset\}$); therefore, the constraint matrix M^P is entirely expressed in \mathcal{AT}_t^+ . Finally, P is projectable, and is therefore equivalent to its projection $\nabla(P)$. $\nabla(P)$, by definition, verifies the following: $(\forall i, j \leq n)((M^{\nabla(P)})_{ji} = \bigcap_{k \leq n} [\nabla^1((M^P)_{ijk}) \cap \nabla^2((M^P)_{kij}) \cap \nabla^3((M^P)_{ikj})])$. Because each of the projections $\nabla^1(R)$, $\nabla^2(R)$ and $\nabla^3(R)$ of any \mathcal{CYC}_t relation R in \mathcal{AT}_t^+ is either a \mathcal{CYC}_b atom or the relation BIN, the \mathcal{CYC}_b -CSP $\nabla(P)$ is entirely expressed in \mathcal{AT}_b^+ .

10 Related work

We compare our approach to cyclic ordering of 2D orientations with the most closely related research in the literature.

10.1 The CYCORD theory

The CYCORD theory [27, 31, 32] expresses cyclic ordering of 2D orientations; it contains only one relation, namely the relation cyc we have already mentioned and translated into the $C\mathcal{YC}_t$ RA (see Example 2(3)). The main disadvantage of the theory is that real applications generally need to represent finer knowledge than just what could be called, as we saw in Example 2(3), strict betweenness in a clockwise direction.

10.2 Representation of a panorama

In [22], Levitt and Lawton discussed QUALNAV, a qualitative landmark navigation system for mobile robots. One feature of the system is the representation of the information about the order of landmarks as seen by the visual sensor of a mobile robot. Such information provides the panorama of the robot with respect to the visible landmarks.

Figure 16 illustrates the panorama of an object S with respect to five reference objects (landmarks) A, B, C, D, E in Schlieder's system [33] (page 527).



Figure 16: The panorama of a location.

The panorama is described by the total cyclic order, in a clockwise direction, of the five directed lines (SA), (SB), (SC), (SD), (SE), and the directed lines which are opposite to them, namely (Sa), (Sb), (Sc), (Sd), (Se): (SA)-(Sc)-(Sd)-(Se)-(Sa)-(SC)-(SD)-(Sb)-(SE). By using the RA \mathcal{CVC}_b , only the five lines joining S to the landmarks are needed to describe the panorama: $\{r((SB), (SA)), r((SC), (SB)), r((SD), (SB)), r((SD), (SC)), l((SE), (SB)), l((SE), (SA))\}$; using the RA \mathcal{CVC}_t , the description can be given as a 2-relation set: $\{rll((SA), (SB), (SE)), rrr((SB), (SC), (SD))\}$.

Schlieder's system makes an implicit assumption, which is that the object to be localised (i.e., S) is not on any of the lines joining pairs of the reference objects – such a fact cannot be represented within his system. This assumption can be made explicit (or indeed could be explicitly contradicted) in the RA $C\mathcal{YC}_b$ representation of the problem (the relations e(qual) and o(pposite) can be used to describe object S being on a line joining two reference objects). Note that Schlieder does not describe the important task of how to reason about a panorama description.

10.3 Cardinal direction models and relative orientation models

Frank's models of cardinal directions in 2D [8, 9] are illustrated in Figure 17. They use a partition of the plane into regions determined by lines passing through a reference object, say S. Depending on the region a point P belongs to, the position of P relative to S is North, North-East, East, South-East, South, South-West, West, North-West, or Equal. Each of the two models can thus be seen as a binary RA, with nine atoms. Both use a global, West-East/South-North, reference system. The projection-based model has been assessed as being cognitively more plausible [8, 9] (cognitive plausibility of spatial orientation models are discussed in [11, 12]), and its computational properties have been studied by Ligozat [24]. In particular, Ligozat made use of tractability results



Figure 17: Frank's cone-shaped (left) and projection-based (right) models of cardinal directions.

known for Allen's interval algebra [1] and Vilain and Kautz's point algebra [39] to find a maximal tractable subset including all atoms (maximal in the sense that adding any other relation to the subset leads to an NP-hard subset). The drawback of Frank's models is that they use a global reference system.

The RA \mathcal{CYC}_b we have presented can be used for the representation of relative orientation knowledge about a configuration of 2D points. Such knowledge would contain for pairs (A, B) of objects in the configuration the position of (the primary object) B relative to (the reference object) A, as viewed from a global point of view, say S: B is on line (SA) on the same side of S as A, to the left of A, on line (SA) on the side of S opposite to that of A, or to the right of A. The drawback here is that the point of view is global.

Thus the common points of Frank's models and our RA \mathcal{CYC}_b are (1) the use of a global concept (a global reference system in the former case, a global viewpoint in the latter); and (2) the representation of knowledge as binary relations describing a primary object relative to a reference object. Their resepctive expressive powers are however incomparable.

A well-known model of relative orientation of 2D points is the Double-Cross calculus defined by Freksa [11], and developed further by Zimmermann and Freksa [40]. The calculus can be looked at as a ternary RA, with 15 atoms corresponding to the 15 regions of a specific partition of the plane determined by a parent object, say A, and a reference object, say B (Figure 18(d)). The partition is based on the following:

- the left/straight/right dichotomy determined by an observer placed at the parent object and looking in the direction of the reference object (Figure 18(a));
- 2. the front/neutral/back dichotomy determined by the same observer (Figure 18(b)); and



Figure 18: The partition of the universe of 2D positions on which is based the Double-Cross calculus in [11, 40].

3. the similar front/neutral/back dichotomy obtained when we swap the roles of the parent object and the reference object (Figure 18(c)).

Combining the three dichotomies (a), (b) and (c) of Figure 18 leads to the partition of the universe of 2D positions on which is based the calculus in [11, 40] (Figure 18(d)).

Our RA \mathcal{CYC}_t can be used for the description of a configuration of 2D points as viewed from a global viewpoint; the model is thus more suited for a panoramalike description. Freksa's calculus, on the other hand, is more suited for the description of a configuration of 2D points (a spatial scene) relative to one another. The two calculi are thus incomparable in terms of expressive power. We have shown that the atoms of \mathcal{CYC}_t form a tractable subset, from which derives a complete solution search procedure for a general problem expressed in the RA. For the Double-Cross calculus, however, no tractable procedure for the subset of all atoms is known.

11 Future work

There has been much work on Allen's interval algebra [1]. For instance:

- The algebra has been shown to be NP-complete [39]. This gave rise to considerable work on tractable subsets of the algebra (see, for instance, [3, 30, 36]); the most important is certainly Nebel and Bürckert's ORD-Horn subclass [30], shown by the authors to be the unique maximal subset among all tractable subsets containing all 13 atomic relations.
- The concept of a conceptual neighbourhood, which is closely related to tractable reasoning, has been extensively investigated for the algebra [10, 11, 23].

Most of this work could be adapted to the two RAs of 2D orientations we have defined. Notice, however, that from corollaries 4 and 5 follows that a tractable

subset of either RA including all atoms cannot include the universal relation, and vice-versa. In terms of expressiveness, the minimal condition for a subset of an RA to be useful is to include all atoms as well as the universal relation; this can be justified thus: (1) it is important for real applications to be offered the possibility of expressing complete information, which is made possible only if all atoms are present; and (2) it is important as well for real applications to be offered the possibility of providing no information on some tuples of the manipulated objects, which is made possible only if the universal relation is present. In the light of these comments, we are committed to face intractability if what we want is to get expressively useful subsets of either RA.

We have provided for the ternary RA a polynomial constraint propagation procedure, which is incomplete in the general case (the RA has been shown to be NP-complete), but still complete for a subset including all atoms. Problems corresponding to actual data (or most randomly generated data) may not lie in the subset. As a consequence, it would be interesting to study the behaviour of a general solution search algorithm, such as the one we have provided (which is exponential in the general case, but solves any problem expressed in the RA), on actual or most randomly generated instances. Again, extensive work on this issue has been done for CSPs of Allen's interval relations [19, 21, 29, 38].

The RAs we have presented do not take into account the *front/neutral/back* dichotomy determined by an observer placed at the point of view and looking in the direction of the reference object; i.e., the dichotomy corresponding to the partition of the plane into the half-plane consisting of the front of the observer, the half-plane consisting of the back of the observer, and the borderline between the two half-planes. Augmenting the binary RA with this feature would lead to eight atoms (equal, left-front, left, left-back, opposite, right-back, right and right-front). The corresponding ternary RA we would lead to has 80 atoms, which can be enumerated by appropriately refining the illustrations of the $C \mathcal{VC}_t$ atoms depicted in Figure 8. For instance, refining the leftmost configuration of the top row in Figure 8 leads to five configurations (see Figure 19). We plan to investigate the computational properties of this finer-grained calculus.

One of the biggest challenges for qualitative spatial reasoning is the integration of qualitative distance and qualitative orientation. A formalism with such a characteristic would, for instance, allow for the representation of natural language descriptions such as "B is closer than, and to left of, A" (B lies within an appropriate sector of the disc centred at the speaker's location, say S, and of radius SA). This challenge has been discussed by many authors [8, 9, 12], and one recent and promising work addressing the issue is [25].

Finally, a calculus of 3D orientations, similar to the ternary RA of 2D orientations we have presented, might be developed.



Figure 19: (Top) The $C \mathcal{YC}_t$ atom lrl, and (Bottom) its refinement resulting from adding the *front/neutral/back* dichotomy to the relation algebras we have presented.

12 Summary

We have provided a new approach to cyclic ordering of 2D orientations, consisting of a relation algebra (RA) whose universe is a set of ternary relations. We have investigated for the RA several algorithmic and computational properties; in particular:

- 1. We have provided a constraint propagation procedure achieving strong 4consistency for a CSP expressed in the RA; and shown that the procedure is polynomial, and complete for a subset including all atoms.
- 2. We have shown that a subset expressing only information on parallel orientations is NP-complete.
- 3. We have shown that provided that a subset S of $C \mathcal{YC}_t$ includes two specific elements, deciding consistency for a CSP expressed in the closure of S under the different operations of the RA can be polynomially reduced to deciding consistency of a CSP expressed in S.
- 4. From the previous result, we have derived that the set obtained by adding the universal relation to the set of all atoms of the RA is NP-complete.
- 5. From the previous result, we have derived that a much less expressive RA, whose universe is a set of binary relations on 2D orientations, is already NP-complete.

We have discussed briefly how this work could be extended, and pointed out to work done so far for other formalisms, such as tractability issues for Allen's algebra of temporal intervals [1], that could be adapted to the two relation algebras we have presented.

References

- [1] J F Allen. Maintaining knowledge about temporal intervals. Communications of the Association for Computing Machinery, 26(11):832-843, 1983.
- [2] B Bennett, A Isli, and A G Cohn. When does a composition table provide a complete and tractable proof procedure for a relational constraint language? In Proceedings of the IJCAI workshop on Spatial and Temporal Reasoning, Nagoya, Japan, 1997.
- [3] C Bessière, A Isli, and G Ligozat. Global consistency in interval algebra networks: tractable subclasses. In *Proceedings ECAI*, pages 3–7, Budapest, Hungary, 1996.
- [4] V Chvátal. Linear Programming. W.H. Freeman and Company, New York, 1983.
- [5] A G Cohn. Qualitative spatial representation and reasoning techniques. In Proceedings KI: German Annual Conference on Artificial Intelligence, volume 1303 of Lecture Notes in Artificial Intelligence, pages 1–30. Springer-Verlag, 1997.
- [6] R Dechter. From local to global consistency. Artificial Intelligence, 55:87-107, 1992.
- [7] G Edwards, G Ligozat, A Gryl, L Fraczak, B Moulin, and C M Gold. A voronoi-based pivot representation of spatial concepts and its application to route descriptions expressed in natural language. In *Proceedings of the International Conference on Spatial Data Handling*, Delft, The Netherlands, 1996.
- [8] A U Frank. Qualitative spatial reasoning about distance and directions in geographic space. Journal of Visual Languages and Computing, 3:343–373, 1991.
- [9] A U Frank. Qualitative spatial reasoning with cardinal directions. In Proceedings of the Austrian Conference on Artificial Intelligence, pages 157– 167, Vienna, 1991.
- [10] C Freksa. Temporal reasoning based on semi-intervals. Artificial Intelligence, 54:199-227, 1992.

- [11] C Freksa. Using orientation information for qualitative spatial reasoning. In A U Frank, I Campari, and U Formentini, editors, Proceedings of International Conference on Theories and Methods of Spatio-Temporal Reasoning in Geographic Space, Berlin, 1992. Springer-verlag.
- [12] C Freksa and K Zimmermann. On the utilization of spatial structures for cognitively plausible and efficient reasoning. In Proceedings of the 1992 IEEE International Conference on Systems, Man and Cybernetics, 1992.
- [13] E C Freuder. Synthesizing constraint expressions. Communications of the Association for Computing Machinery, 21:958–966, 1978.
- [14] E C Freuder. A sufficient condition for backtrack-free search. Journal of the Association for Computing Machinery, 29:24-32, 1982.
- [15] Z Galil and N Megiddo. Cyclic ordering is NP-complete. Theoretical Computer Science, (5):199-182, 1977.
- [16] M R Garey and D S Johnson. Computers and intractability. Freeman, New York, 1979.
- [17] A Gryl, M Denis, and G Ligozat. La description d'itinéraires: éléments pour un générateur en langage naturel. In Proceedings of XXVIème Congrès International de Psychologie, Montreal, 1996.
- [18] D Hernández. Relative representation of spatial knowledge: the 2-d case. In Cognitive and Linguistic Aspects of Geographic Space, Nato Advanced Studies Institute. Kluwer, 1991.
- [19] A Isli and H Bennaceur. Qualitative interval networks: combining path consistency and circuit consistency in the search for a solution. In Proceedings of the International Workshop of Temporal Representation and Reasoning (TIME-96), Key West, Florida, 1996.
- [20] P Ladkin and R Maddux. On binary constraint problems. Journal of the Association for Computing Machinery, 41(3):435-469, 1994.
- [21] P Ladkin and A Reinefeld. Effective solution of qualitative constraint problems. Artificial Intelligence, 57:105–124, 1992.
- [22] T S Levitt and D T Lawton. Qualitative navigation for mobile robots. Artificial Intelligence, 44(2):305-360, 1990.
- [23] G Ligozat. Towards a general characterization of conceptual neighbourhoods in temporal and spatial reasoning. In F D Anger and R Loganantharah, editors, Proceedings AAAI-94 Workshop on Spatial and Temporal Reasoning, 1994.

- [24] G Ligozat. Reasoning about cardinal directions. Journal of Visual Languages and Computing, 9(1):23-44, 1998.
- [25] J Liu. A method of spatial reasonig based on qualitative trigonometry. Artificial Intelligence, 98:137-168, 1998.
- [26] A K Mackworth. Consistency in networks of relations. Artificial Intelligence, 8:99-118, 1977.
- [27] N Megiddo. Partial and complete cyclic orders. Bull. Am. Math. Soc., 82:274-276, 1976.
- [28] U Montanari. Networks of constraints: Fundamental properties and applications to picture processing. *Information Sciences*, 7:95–132, 1974.
- [29] B Nebel. Solving hard qualitative temporal reasoning problems: Evaluating the efficiency of using the ord-horn class. CONSTRAINTS, 1(3):175–190, 1997.
- [30] B Nebel and H-J Bürckert. Reasoning about temporal relations: a maximal tractable subset of Allen's interval algebra. *Journal of the Association for Computing Machinery*, 42(1):43-66, January 1995.
- [31] R Röhrig. A theory for qualitative spatial reasoning based on order relations. In *Proceedings AAAI*, Seattle, 1994.
- [32] R Röhrig. Representation and processing of qualitative orientation knowledge. In Proceedings KI: German Annual Conference on Artificial Intelligence, volume 1303 of Lecture Notes in Artificial Intelligence, pages 219– 230. Springer-Verlag, 1997.
- [33] C Schlieder. Representing visible locations for qualitative navigation. In *Qualitative Reasoning and Decision Technologies: CIMNE*, 1993.
- [34] C Schlieder. Qualitative shape representation. In A Frank, editor, Spatial conceptual models for geographic objects with undetermined boundaries. London: Taylor and Francis, 1994.
- [35] A Tarski. On the calculus of relations. Journal of Symbolic Logic, 6:73–89, 1941.
- [36] P van Beek. Reasoning about qualitative temporal information. Artificial Intelligence, 58:297-326, 1992.
- [37] P van Beek and R Cohen. Exact and approximate reasoning about temporal relations. *Computational Intelligence*, 6:132–144, 1990.

- [38] P van Beek and D M Manchak. The design and experimental analysis of algorithms for temporal reasoning. Journal of Artificial Intelligence Research, 4:1–18, 1996.
- [39] M B Vilain and H Kautz. Constraint propagation algorithms for temporal reasoning. In *Proceedings AAAI-86*, Philadelphia, August 1986.
- [40] K Zimmermann and C Freksa. Qualitative spatial reasoning using orientation, distance, and path knowledge. *Applied Intelligence*, 6:49–58, 1996.

A Verifying the RA properties for CYC_b

For the RA \mathcal{CYC}_t , presented in Section 6, which is strictly more expressive than \mathcal{CYC}_b , we will verify all the RA properties. Thus, it is not necessary to verify them for \mathcal{CYC}_b . We observe, however, that one has to check, at the level of atoms, that the results of applying the operations of converse and composition to the different atoms are correct; in other words, we have to check for \mathcal{CYC}_b the following:

- 1. the entries of the converse table: the entry Conv(b) on row b of the converse table must be equal to the converse b^{\smile} of b, i.e., we must have $Conv(b) = b^{\smile} = \{(x, y) : (y, x) \in b\}$; and
- 2. the entries of the composition table: the entry $T(b_1, b_2)$ on row b_1 and column b_2 of the composition table must be equal to the composition, $b_1 \circ b_2$, of atoms b_1 and b_2 , i.e., we must have $T(b_1, b_2) = b_1 \circ b_2 = \{(x, y) : (\exists z)(b_1(x, z) \land b_2(z, y))\}.$

We will show that this is indeed the case for \mathcal{CYC}_b . As we saw before, given two orientations X and Y:

$$e(Y,X) \text{ iff } (X,Y) \in \{0\}$$

$$\tag{4}$$

$$l(Y,X) \text{ iff } (X,Y) \in (0,\pi) \tag{5}$$

$$o(Y, X) \text{ iff } (X, Y) \in \{\pi\}$$

$$(6)$$

$$r(Y,X) \text{ iff } (X,Y) \in (\pi,2\pi) \tag{7}$$

In other words, the atoms e, l, o, r correspond, respectively, to the convex subsets $\{0\}, (0, \pi), \{\pi\}, (\pi, 2\pi)$ of $[0, 2\pi)$.

Checking the entries of the converse table: By definition, $e^{\frown} = \{(x, y) : (y, x) \in e\}, \overline{l^{\frown} = \{(x, y) : (y, x) \in l\}}, \overline{o^{\frown} = \{(x, y) : (y, x) \in o\}}, \overline{r^{\frown} = \{(x, y) : (y, x) \in e\}}, \overline{r^{\frown} = \{(x, y) : (y, x) \in e\}}, \overline{r^{\frown} = \{(x, y) : (y, x) \in e\}}, \overline{r^{\frown} = \{(x, y) : (x, y) \in d\}}, \overline{r^{\frown} = \{(x, y) : (x, y) \in \{0\}}, \overline{r^{\frown} = \{(x, y) : (x, y) \in \{0\}}, \overline{r^{\frown} = \{(x, y) : (x, y) \in \{0\}}, \overline{r^{\frown} = \{(x, y) : (x, y) \in \{0\}}, \overline{r^{\frown} = \{(x, y) : (x, y) \in \{0\}}, \overline{r^{\frown} = \{(x, y) : (x, y) \in \{\pi\}}, \overline{r^{\frown} = \{(x, y) : (x, y) \in (\pi, 2\pi)\}}.$ The assertions $(x, y) \in \{0\}, (x, y) \in \{0\}, (x, y) \in \{\pi\}, (y, x) \in \{\pi\}, (y, x) \in \{0, \pi\}, we get: e^{\frown} = \{(x, y) : (y, x) \in \{0\}\}, \overline{l^{\frown} = \{(x, y) : (y, x) \in \{0\}\}}, \overline$



Figure 20: Illustration of the proof of Theorem 10.

 $\begin{array}{l} \{(x,y):(y,x)\in(\pi,2\pi)\}, o^{\smile}=\{(x,y):(y,x)\in\{\pi\}\}, r^{\smile}=\{(x,y):(y,x)\in(0,\pi)\}. \text{ Using again the equivalences 4-7, we get that the converse table records the exact converses of the atoms: } e^{\smile}=\{(x,y):e(x,y)\}=e, l^{\smile}=\{(x,y):r(x,y)\}=r, o^{\smile}=\{(x,y):o(x,y)\}=o, r^{\smile}=\{(x,y):l(x,y)\}=l. \end{array}$

Checking the entries of the composition table: In order to check that $T(b_1, b_2) = b_1 \circ b_2$, it is sufficient to use the following sound inference rule, in which A and B denote convex subsets of $[0, 2\pi)$, and size(X) is the maximum of all y - x for $x, y \in X$:

$$[(X, Z) \in A \land (Z, Y) \in B \land size(A) < \pi \land size(B) < \pi] \Rightarrow (X, Y) \in A +_{s} B]$$
(8)

where $+_s$ is set addition (composition): $A +_s B = \{c : (\exists a \in A, \exists b \in B) (c = a + b)\}$. We claim that the inference rule is 3-complete for $A, B \in \{\{0\}, (0, \pi), \{\pi\}, (\pi, 2\pi)\}$; i.e., for any such A and B, we have the following: $(\forall X, Y)[(X, Y) \in A +_s B \Rightarrow (\exists Z)((X, Z) \in A \land (Z, Y) \in B)]$.

Theorem 10 The inference rule (8) is 3-complete for $A, B \in \{\{0\}, (0, \pi), \{\pi\}, (\pi, 2\pi)\}$.

Proof. We proceed by enumerating all possible cases. Cases (1) and (6) in the enumeration are illustrated in Figure 20:

- 1. If $A = \{0\}$ then $A +_s B = B$. For all X, Y such that $(X, Y) \in B$, if we take Z = X then $(X, Z) = (X, X) = 0 \in A$ and $(Z, Y) = (X, Y) \in B$ (see Figure 20(a)).
- 2. If $A = \{\pi\}$ then $A +_s B = \{\pi + \beta : \beta \in B\}$. Let X, Y be such that $(X, Y) \in A +_s B$: $(\exists \beta \in B)((X, Y) = \pi + \beta)$. We take $Z = X + \pi$; then from $(X, Y) = \pi + \beta$ and $Z = \pi + X$, we infer that $(X, Z) = \pi \in A$ and $(Z, Y) = \beta \in B$.
- 3. If $B = \{0\}$ then $A +_s B = A$. For all X, Y such that $(X, Y) \in A$, if we take Z = Y then $(X, Z) = (X, Y) \in A$ and $(Z, Y) = (Y, Y) = 0 \in B$.

- 4. If $B = \{\pi\}$ then $A +_s B = \{\alpha + \pi : \alpha \in A\}$. Let X, Y be such that $(X, Y) \in A +_s B$: $(\exists \alpha \in A)((X, Y) = \alpha + \pi)$. We take $Z = Y + \pi$; thus $Y = Z + \pi$, from which we can infer $(Z, Y) = \pi \in B$. From $(X, Y) = \alpha + \pi$ and $(Y, Z) = \pi$, we infer $(X, Z) = \alpha \in A$.
- 5. If $A = (0, \pi)$ and $B = (0, \pi)$ then $A +_s B = \{\alpha + \beta : \alpha \in (0, \pi), \beta \in (0, \pi)\} = (0, 2\pi)$. For all X and Y such that $(X, Y) \in (0, 2\pi)$, we can find Z such that $(X, Z) \in (0, \pi)$ and $(Z, Y) \in (0, \pi)$: take Z in such a way that (X, Z) = (Z, Y) (i.e., Z is the bisector of (X, Y)).
- 6. If $A = (0, \pi)$ and $B = (\pi, 2\pi)$ then $A +_s B = \{\alpha + \beta : \alpha \in (0, \pi), \beta \in (\pi, 2\pi)\} = (\pi, 3\pi) \equiv [0, \pi) \cup (\pi, 2\pi)$. For all X and Y such that $(X, Y) \in [0, \pi) \cup (\pi, 2\pi)$, we can find Z such that $(X, Z) \in (0, \pi)$ and $(Z, Y) \in (\pi, 2\pi)$: if $(X, Y) \in (\pi, 2\pi)$ (see Figure 20(b)), take Z in such a way that (X, Z) = (Z, Y'), where Y' is the orientation opposite to Y (i.e., $(Y, Y') = \pi$); if (X, Y) = 0 (see Figure 20(c)), take Z in such a way that $(X, Z) = \pi/2$; and if $(X, Y) \in (0, \pi)$ (see Figure 20(d)), take Z in such a way that (Y, Z) = (Z, X'), where X' is the orientation opposite to X (i.e., $(X, X') = \pi$).
- 7. If $A = (\pi, 2\pi)$ and $B = (0, \pi)$ then $A +_s B = \{\alpha + \beta : \alpha \in (\pi, 2\pi), \beta \in (0, \pi)\} = (\pi, 3\pi) \equiv [0, \pi) \cup (\pi, 2\pi)$. For all X and Y such that $(X, Y) \in (\pi, 3\pi)$, we can find Z such that $(X, Z) \in (\pi, 2\pi)$ and $(Z, Y) \in (0, \pi)$: if $(X, Y) \in (\pi, 2\pi)$, take Z in such a way that (X', Z) = (Z, Y) (i.e., Z is the bisector of (X', Y)), where X' is the orientation opposite to X (i.e., X' is such that $(X, X') = \pi$); if (X, Y) = 0, take Z in such a way that $(Y', Z) = 3\pi/2$; and if $(X, Y) \in (0, \pi)$, take Z in such a way that (Y', Z) = (Z, X) (i.e., Z is the bisector of (Y', X)), where Y' is the orientation opposite to Y (i.e., Y' is such that $(Y, Y') = \pi$).
- 8. If $A = (\pi, 2\pi)$ and $B = (\pi, 2\pi)$ then $A +_s B = \{\alpha + \beta : \alpha \in (\pi, 2\pi), \beta \in (\pi, 2\pi)\} = (2\pi, 4\pi) \equiv (0, 2\pi)$. For all X and Y such that $(X, Y) \in (0, 2\pi)$, we can find Z such that $(X, Z) \in (\pi, 2\pi)$ and $(Z, Y) \in (\pi, 2\pi)$: if $(X, Y) \in (\pi, 2\pi)$, take Z in such a way that (Y, Z) = (Z, X) (i.e., Z is the bisector of (Y, X)); if $(X, Y) = \pi$, take Z in such a way that (X', Z) = (Z, Y') (i.e., Z is the bisector of $(X, Y) \in (0, \pi)$, take Z in such a way that (X', Z) = (Z, Y') (i.e., Z is the bisector of (X', Y')), where X' is the orientation opposite to X (i.e., $(X, X') = \pi$) and Y' is the orientation opposite to Y (i.e., $(Y, Y') = \pi$).

B Verifying the RA properties for an atomic ternary RA

1. $\frac{(R \circ S) \circ T = R \circ (S \circ T)?}{(a) \text{ Let } (a, b, c) \in (R \circ S) \circ T. \text{ Thus } (\exists d)((a, b, d) \in (R \circ S) \land (a, d, c) \in T).$

 $(a, b, d) \in (R \circ S)$ implies $(\exists e)((a, b, e) \in R \land (a, e, d) \in S)$. From $(a, e, d) \in S$ and $(a, d, c) \in T$, we infer $(a, e, c) \in (S \circ T)$. From $(a, b, e) \in R$ and $(a, e, c) \in (S \circ T)$, we infer $(a, b, c) \in R \circ (S \circ T)$. Therefore $(R \circ S) \circ T \subseteq R \circ (S \circ T)$.

(b) Now let $(a, b, c) \in R \circ (S \circ T)$. Thus $(\exists d)((a, b, d) \in R \land (a, d, c) \in (S \circ T))$. $(a, d, c) \in (S \circ T)$ implies $(\exists e)((a, d, e) \in S \land (a, e, c) \in T)$. From $(a, b, d) \in R$ and $(a, d, e) \in S$, we infer $(a, b, e) \in (R \circ S)$. From $(a, b, e) \in (R \circ S)$ and $(a, e, c) \in T$, we infer $(a, b, c) \in (R \circ S) \circ T$. Therefore $(R \circ S) \circ T \supseteq R \circ (S \circ T)$.

- 2. $\frac{(R \cup S) \circ T = (R \circ T) \cup (S \circ T)?}{(R \cup S) \circ T = \{(a, b, c) : (\exists d)((a, b, d) \in (R \cup S) \land (a, d, c) \in T)\} = \{(a, b, c) : (\exists d)([(a, b, d) \in R \lor (a, b, d) \in S]) \land (a, d, c) \in T\} = \{(a, b, c) : (\exists d)((a, b, d) \in R \land (a, d, c) \in T)\} \cup \{(a, b, c) : (\exists d)((a, b, d) \in S \land (a, d, c) \in T)\} = \{(R \circ T) \cup (S \circ T). \blacksquare$
- 3. $\underline{R \circ \mathcal{I} = \mathcal{I} \circ R = R?}$

We prove this for $\mathcal{I} = \mathcal{I}_{U}^{t_{23}} = \{(a, b, b) : a, b \in U\}$. The reason for this is that the identity element of the atomic ternary RA of our interest, \mathcal{CYC}_t , is $\mathcal{I}_{2D\mathcal{O}}^{t_{23}}$ $(U = 2D\mathcal{O})$. $R \circ \mathcal{I} = \{(a, b, c) : (\exists d)((a, b, d) \in R \land (a, d, c) \in \mathcal{I})\}$. But $(a, d, c) \in \mathcal{I}$ implies (d = c); thus $R \circ \mathcal{I} = \{(a, b, c) : (a, b, c) \in R \land (a, c, c) \in \mathcal{I}\}$. Since $(\forall a, c)((a, c, c) \in \mathcal{I})$, we infer $R \circ \mathcal{I} = \{(a, b, c) : (a, b, c) \in R\} = R$. On

 $(\forall a, c)((a, c, c) \in \mathcal{I})$, we infer $R \circ \mathcal{I} = \{(a, b, c) : (a, b, c) \in R\} = R$. On the other hand, $\mathcal{I} \circ R = \{(a, b, c) : (\exists d)((a, b, d) \in \mathcal{I} \land (a, d, c) \in R)\}$. But $(a, b, d) \in \mathcal{I}$ implies (d = b); thus $\mathcal{I} \circ R = \{(a, b, c) : (a, b, b) \in \mathcal{I} \land (a, b, c) \in R\}$. Since $(\forall a, b)((a, b, b) \in \mathcal{I})$, we infer $\mathcal{I} \circ R = \{(a, b, c) : (a, b, c) \in R\} = R$.

- 4. $(R^{\smile})^{\smile} = R$? $(R^{\smile})^{\smile} = \{(a, b, c) : (a, c, b) \in R^{\smile}\}$. But $(a, c, b) \in R^{\smile}$ is equivalent to $(a, b, c) \in R$. Therefore $(R^{\smile})^{\smile} = \{(a, b, c) : (a, b, c) \in R\} = R$. ■
- 5. $\frac{(R \cup S)^{\smile} = R^{\smile} \cup S^{\frown}?}{(R \cup S)^{\smile} = \{(a, b, c) : (a, c, b) \in R \cup S\}} = \{(a, b, c) : (a, c, b) \in R \lor (a, c, b) \in S\} = \{(a, b, c) : (a, c, b) \in R\} \cup \{(a, b, c) : (a, c, b) \in S\} = R^{\smile} \cup S^{\smile}. \blacksquare$
- 6. $\frac{(R \circ S)^{\smile} = S^{\smile} \circ R^{\frown}?}{(R \circ S)^{\smile} = \{(a, b, c) : (a, c, b) \in R \circ S\}} = \{(a, b, c) : (\exists d)((a, c, d) \in R \land (a, d, b) \in S)\} = \{(a, b, c) : (\exists d)((a, d, c) \in R^{\smile} \land (a, b, d) \in S^{\frown})\} = \{(a, b, c) : (\exists d)((a, b, d) \in S^{\frown} \land (a, d, c) \in R^{\frown})\} = S^{\frown} \circ R^{\frown}. \blacksquare$
- 7. $\frac{R^{\smile} \circ \overline{R \circ S} \cap S = \emptyset?}{\text{Let } (a, b, c) \in \overline{R^{\smile} \circ S} \circ \overline{R \circ S}}.$ Thus $(\exists d)((a, b, d) \in \overline{R^{\smile}} \land (a, d, c) \in \overline{R \circ S}).$ $(a, d, c) \in \overline{R \circ S}$ is equivalent to $(a, d, c) \notin R \circ S$, which in turn implies $(\forall e)((a, d, e) \notin R \lor (a, e, c) \notin S).$ Now consider the special case e = b: from $(a, b, d) \in R^{\smile}$, we derive $(a, d, b) \in R$; thus $(a, b, c) \notin S$.
 - 54

- 8. $\frac{((R^{\frown})^{\frown})^{\frown} = R?}{((R^{\frown})^{\frown})^{\frown} = \{(a, b, c) : (c, a, b) \in (R^{\frown})^{\frown}\}}.$ But $(c, a, b) \in (R^{\frown})^{\frown}$ is equivalent to $(b, c, a) \in R^{\frown}$, which in turn is equivalent to $(a, b, c) \in R$. Therefore $((R^{\frown})^{\frown})^{\frown} = \{(a, b, c) : (a, b, c) \in R\} = R$.
- $\begin{array}{l} 9. \ \underline{(R \cup S)^{\frown} = R^{\frown} \cup S^{\frown}?} \\ \hline (R \cup S)^{\frown} = \{(a, b, c) : (c, a, b) \in R \cup S\} = \{(a, b, c) : (c, a, b) \in R \lor (c, a, b) \in S\} \\ S\} = \{(a, b, c) : (c, a, b) \in R\} \cup \{(a, b, c) : (c, a, b) \in S\} = R^{\frown} \cup S^{\frown}. \end{array}$
- 10. Checking the entries of the different tables: Similarly to \mathcal{CYC}_b , we have to check that the converse table, the rotation table and the composition tables of \mathcal{CYC}_t record the exact converses, the exact rotations and the exact compositions of the atoms.

<u>The converse table and the rotation table:</u> From the fact that the \mathcal{CYC}_b converse table records the exact converses of the atoms, we derive straightforwardly that the converse table and the rotation table of \mathcal{CYC}_t record the exact converses and the exact rotations of the atoms. We illustrate this with the atom *lrr*. By definition, $(lrr)^{\smile} = \{(x, y, z) : (x, z, y) \in lrr\}.$ Applying the definition of a \mathcal{CYC}_t atom, we get: $(lrr)^{\smile} = \{(x, y, z) :$ $(z,x) \in l \land (y,z) \in r \land (y,x) \in r$. Reordering the elements of the conjunction $(z,x) \in l \land (y,z) \in r \land (y,x) \in r$, we get: $(lrr)^{\smile} = \{(x,y,z):$ $(y,x) \in r \land (y,z) \in r \land (z,x) \in l$. Thanks to the fact that the \mathcal{CYC}_b converse table records the exact converses of the atoms, we derive that $(y,z) \in r$ iff $(z,y) \in l$, from which we get: $(lrr)^{\smile} = \{(x,y,z) : (y,x) \in l\}$ $r \wedge (z, y) \in l \wedge (z, x) \in l$. Now the set $\{(x, y, z) : (y, x) \in r \wedge (z, y) \in v\}$ $l \wedge (z, x) \in l$ corresponds exactly the \mathcal{CYC}_t atom rll, which implies that $(lrr)^{\sim} = rll$. By definition of the rotation operation, we get: $(lrr)^{\sim} =$ $\{(x, y, z) : (z, x, y) \in lrr\}$. Using the definition of a \mathcal{CYC}_t atom, we get: $(lrr)^{\frown} = \{(x, y, z) : (x, z) \in l \land (y, x) \in r \land (y, z) \in r\}.$ Reordering the elements of the conjunction $(x, z) \in l \land (y, x) \in r \land (y, z) \in r$, we get: $(lrr)^{\frown} = \{(x, y, z) : (y, x) \in r \land (y, z) \in r \land (x, z) \in l\}$. Thanks, again, to the fact that the \mathcal{CYC}_b converse table records the exact converses of the atoms, we get that the assertions $(y, z) \in r$ and $(x, z) \in l$ are equivalent, respectively, to $(z, y) \in l$ and $(z, x) \in r$, which implies: $(lrr)^{\frown} = \{(x, y, z) : (y, x) \in r \land (z, y) \in l \land (z, x) \in r\}$. Now the set $\{(x, y, z) : (y, x) \in r \land (z, y) \in l \land (z, x) \in r\}$ corresponds exactly to the \mathcal{CYC}_t atom rlr; thus $(lrr)^{\frown} = rlr$.

<u>The composition tables:</u> We say that the \mathcal{CYC}_t composition tables are sound if for any two atoms t_1 and t_2 , it is the case that $T(t_1, t_2) \supseteq t_1 \circ t_2$, where $T(t_1, t_2)$ is the entry on the row labelled with t_1 and the column labelled with t_2 ; if the tables are sound, we say that they are 4-complete if for any two atoms t_1 and t_2 , it is the case that $T(t_1, t_2) \subseteq t_1 \circ t_2$. Soundness implies that if we know that a triple (x, y, w) belongs to $t_1 \circ t_2$, which, by definition, means that we can find z such that $(x, y, z) \in t_1$ and $(x, z, w) \in t_2$, then it must be the case that the triple (x, y, w) also

belongs to the entry $T(t_1, t_2)$; if 4-completeness also holds then the triples (x, y, w) in the relation recorded by an entry correspond exactly to the actual composition of the corresponding atoms. We show how to compute the entries of the composition tables; this will at the same time show 4-completeness of the tables. For this purpose, we consider two atoms $t_1 = b_1 b_2 b_3$ and $t_2 = b'_1 b'_2 b'_3$. As we saw before, due to the fact that the \mathcal{CVC}_b atoms are JEPD, $t_1 \circ t_2 = \emptyset$ if $b_3 \neq b'_1$ (again, refer to Figure 6 for illustration); so we suppose $b_3 = b'_1 = b$, which leads to $t_1 = b_1 b_2 b$, $t_2 = bb'_2b'_3$, and $t_1 \circ t_2 = \{(x, y, w) : (\exists z)((y, x) \in b_1 \land (z, y) \in b_2 \land (z, x) : ($ $b \wedge (w, z) \in b'_2 \wedge (w, x) \in b'_3$. We will need the isomorphim ϕ from $2D\mathcal{O} \times 2D\mathcal{O}$ onto $2D\mathcal{O} \times 2D\mathcal{O}$, defined as follows: $\phi((x,y)) = (x',y)$, where x' the orientation opposite to x, i.e., x' is such that o(x, x'); the isomorphism is extended to subsets of $2D\mathcal{O} \times 2D\mathcal{O}$ in the following natural way: $\phi(S) = \{\phi((x, y)) : (x, y) \in S\}$; for the \mathcal{CVC}_b atoms, which are particular subsets of $2D\mathcal{O} \times 2D\mathcal{O}$, we get $\phi(e) = o, \phi(l) = r, \phi(o) = e, \phi(r) = l$. We proceed by enumerating all possible cases:

- (a) if $b_1 = e$ then $t_1 \circ t_2 = eb'_3b'_3$;
- (b) if $b_2 = e$ then $t_1 \circ t_2 = t_2$;
- (c) if $b'_2 = e$ then $t_1 \circ t_2 = t_1$;
- (d) if $b'_3 = e$ then $t_1 \circ t_2 = b_1(b_1) \ e_i$;
- (e) if $b_1 = o$ then from $(y, x) \in o \land (w, x) \in b'_3$ we get $(w, y) \in \phi(b'_3)$; thus $t_1 \circ t_2 = \{(x, y, w) : (y, x) \in b_1 \land (w, y) \in \phi(b'_3) \land (w, x) \in b'_3\} = b_1\phi(b'_3)b'_3$;
- (f) if $b'_3 = o$ then from $(x, y) \in (b_1)^{\smile} \land (w, x) \in o$ we get $(w, y) \in \phi((b_1)^{\smile})$; thus $t_1 \circ t_2 = \{(x, y, w) : (y, x) \in b_1 \land (w, y) \in \phi((b_1)^{\smile}) \land (w, x) \in b'_3\} = b_1\phi((b_1)^{\smile})b'_3$;
- (g) if $b_2 = o$ then from $(w, z) \in b'_2 \land (z, y) \in o$ we get $(w, y) \in \phi(b'_2)$; thus $t_1 \circ t_2 = \{(x, y, w) : (y, x) \in b_1 \land (w, y) \in \phi(b'_2) \land (w, x) \in b'_3\} = b_1 \phi(b'_2) b'_3$;
- (h) if $b'_2 = o$ then from $(z, y) \in b_2 \land (w, z) \in o$ we get $(w, y) \in \phi(b_2)$; thus $t_1 \circ t_2 = \{(x, y, w) : (y, x) \in b_1 \land (w, y) \in \phi(b_2) \land (w, x) \in b'_3\} = b_1\phi(b_2)b'_3$;
- (i) if b = e then JEPDness of the CYC_b atoms gives b₂ = (b₁)[→] and b'₂ = b'₃; this leads to t₁ ∘ t₂ = Π(b₁, b'₃ ∘ (b₁)[→], b'₃) (4-completeness comes from 3-completess of the CYC_b composition table: each entry records the exact composition of the corresponding CYC_b atoms);
- (j) In a similar way, if b = o then we infer that $b'_3 = \phi(b'_2)$ and $b_1 = \phi((b_2)^{\smile})$; thus $t_1 \circ t_2 = \Pi(b_1, b'_3 \circ (b_1)^{\smile}, b'_3)$. Again, 4-completeness stems from 3-completeness of the \mathcal{CYC}_b composition table.
 - 56



Figure 21: Illustration of 4-completeness of the \mathcal{CYC}_t composition tables.

The remaining cases are those when each of b_1, b_2, b, b'_2, b'_3 belongs to $\{l, r\}$. These cover altogether 32 entries of the composition tables: 16 of these consist of atoms, the other 16 of 3-atom relations. We prove 4-completeness for one 1-atom entry and for one 3-atom entry; the 4-completeness proof for the other entries is similar. We consider the entries T(llr, rll) = lrl and $T(llr, rlr) = \{llr, lor, lrr\}$.

 $\frac{T(llr,rll) = lrl?}{\overline{drl}}$

Consider four orientations x, y, z, w such that $llr(x, y, z) \wedge rll(x, z, w)$. This is illustrated in Figure 21(left). Orientation w is forced to be between-ina-clockwise-direction the orientation opposite to z and orientation x. The illustration clearly indicates that the relation on triple (x, y, w) is lrl. Conversely, consider a configuration of three orientations x, y and w such that lrl(x, y, w). We can always find z such that $llr(x, y, z) \wedge rll(x, z, w)$: for instance, we can take z such that o(z, z'), where z' in turn is such that (w, z') = (z', y) (z' is the bisector of (w, y)). $T(llr, rlr) = \{llr, lor, lrr\}$?

Consider four orientations x, y, z, w such that $llr(x, y, z) \wedge rlr(x, z, w)$. This is illustrated in Figure 21(right). Orientation w is forced to be to the left of, opposite to, or to the right of, y; thus the relation on triple (x, y, w)is $\Pi(l, \{l, o, r\}, r) = \{llr, lor, lrr\}$. Conversely, consider a configuration of three orientations x, y and w such that $\{llr, lor, lrr\}(x, y, w)$. We can always find z such that $llr(x, y, z) \wedge rlr(x, z, w)$: if llr(x, y, w) or lor(x, y, w)then take z such that (x', z) = (z, w), where x' is such that o(x, x'); otherwise, take z such that (x', z) = (z, y'), where x' and y' are such that o(x, x') and o(y, y').